ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Martínez, Luis R.; Jessen, Jonas; Xu, Guo

Working Paper A Glimpse of Freedom: Allied Occupation and Political Resistance in East Germany

IZA Discussion Papers, No. 15606

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Martínez, Luis R.; Jessen, Jonas; Xu, Guo (2022) : A Glimpse of Freedom: Allied Occupation and Political Resistance in East Germany, IZA Discussion Papers, No. 15606, Institute of Labor Economics (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/265827

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 15606

A Glimpse of Freedom: Allied Occupation and Political Resistance in East Germany

Luis R. Martinez ⁽⁷⁾ Jonas Jessen Guo Xu

SEPTEMBER 2022



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 15606

A Glimpse of Freedom: Allied Occupation and Political Resistance in East Germany

Luis R. Martinez ©

University of Chicago

Jonas Jessen European University Viadrina, DIW Berlin and IZA

Guo Xu University of California at Berkeley and NBER

SEPTEMBER 2022

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

ABSTRACT

A Glimpse of Freedom: Allied Occupation and Political Resistance in East Germany^{*}

This paper exploits the idiosyncratic line of contact separating Allied and Soviet troops within East Germany at the end of WWII to study political resistance in a non-democracy. When Nazi Germany surrendered, 40% of what would become the authoritarian German Democratic Republic was initially under Allied control but was ceded to Soviet control less than two months later. Brief Allied exposure increased protests during the major 1953 uprising. We use novel data on the appointment of local mayors and a retrospective survey to argue that even a "glimpse of freedom" can foster civilian opposition to dictatorship.

JEL Classification:	F51, H10, N44, P20
Keywords:	East Germany, political resistance, protest, autocracy, spatial RDD, World War II

Corresponding author: Jonas Jessen

DIW Berlin Mohrenstraße 58 10117 Berlin Germany E-mail: jessen@diw.de

^{*} Mariella Gonzales and Jiaqi Zhang provided excellent research assistance. We would like to thank Peter Buisseret, Ernesto Dal Bo, Steven Durlauf, Christian Ochsner, Aleksandra Peeva, Mehdi Shadmehr, Jacob Shapiro, Nico Voigtländer and Joachim Voth for helpful comments. We thank audiences at DIW Berlin and the Political Economy in the Chicago Area (PECA) workshop for comments and helpful suggestions. We also thank Michael Löderbusch and Florian Lionel Reichert for generously piloting the survey, and Laura Löderbusch for polishing the maps. We are indebted to Holger Kern for kindly sharing data with us and with Jenns Robertson for kindly answering our questions on the THOR dataset. We gratefully acknowledge funding from the Pearson Institute for the Study and Resolution of Global Conicts. This project received IRB exemption from the University of California at Berkeley (2019-10-12584) and the University of Chicago (19-1398). The use of ① signals that the authors' names have been randomized using a uniform distribution. This article will appear in American Economic Journal: Applied Economics.

1 Introduction

Countries experiencing declines in civil liberties and political rights have outnumbered those with improvements for more than a decade. Over 60% of the world population now lives in countries that are less than fully free (Freedom House, 2020). In many of these countries, overt political opposition carries a risk for people's freedom, integrity and lives. Canonical models of regime change predict that people's willingness to rebel increases as their expected gain from regime change becomes clearer or larger (Acemoglu and Robinson, 2005). In practice, however, people living in non-democracies often only have a vague notion of what would happen were the current regime to fall. One-party states and absolute monarchies, like China or Saudi Arabia, leave no formal alternative. In hybrid regimes, like Russia, Turkey or Venezuela, opposition parties and their leaders are often suppressed. Lack of exposure to alternatives to the status quo may thus play an important role in explaining (the lack of) costly political opposition in non-democracies.

In this paper, we show that even a short exposure to better governance and a more disciplined occupying force can increase subsequent resistance to autocratic rule in the early stages of nation-building. Our setting focuses on the consolidation of the German Democratic Republic (GDR), the nation that emerged from the Soviet-administered occupation zone in Germany after World War II. We examine an understudied aspect of the end of the war in Germany. At the time of the Nazi surrender, 40% of the pre-arranged Soviet occupation zone was occupied by troops from the western Allied Expeditionary Force led by the US and the UK. This Allied occupation was very short-lived and Soviet forces were in full control less than two months later.¹ Over the following years, the entire Soviet-controlled area would develop into a highly authoritarian and repressive regime: the GDR, also known as East Germany.

Our empirical analysis exploits the idiosyncratic nature of the *line of contact* separating Allied and Soviet forces at the end of the war *within* East Germany. We leverage historical sources to show that the line emerged through a series of haphazard military decisions made in the very final days of the war as the Allied and Soviet armies fought their way into Germany from opposite directions. As a result, part of the area assigned for Soviet occupation remained under Allied rule for around two months before the Soviet takeover, while the rest

¹We use the term "Allies" throughout the paper to refer to the Western Allies, formally the Allied Expeditionary Force, comprised mostly of military units from the United States and the United Kingdom. We use the term "Grand Alliance" to refer to the larger group of countries, including the USSR, that fought against the Axis powers (Germany, Italy and Japan) in WWII.

experienced Soviet rule throughout. We conduct careful historical and archival research to reconstruct the exact line of contact at the granular municipality level (*Gemeinde*). We then use a spatial regression discontinuity design (RDD) to study the effects of a fleeting exposure to Allied occupation on measures of political resistance to the budding dictatorship in East Germany.

Our main outcome of interest is the incidence of protests during the 1953 uprising in the GDR. This was the only major episode of political unrest to take place in the country until the events of 1989/90 that led to democratization and German reunification. The uprising would have arguably caused the demise of the communist dictatorship in East Germany, had it not been for Soviet military intervention (Grieder, 2012). We show that municipalities that were initially occupied by Allied forces were approximately 15 percentage points (pp) more likely to experience protests during the 1953 uprising. This is a large effect, corresponding to 68% of the sample mean.

We validate our findings with a standard battery of tests. First, we provide evidence of balance in pre-determined covariates at the line of contact. Second, we verify that our results hold up if we include border-segment fixed effects, which constrain the comparison to municipalities in direct vicinity, or if we add a rich set of controls. Third, our results are also robust to different functional forms for the running variable or different bandwidths, including the optimal bandwidth following Calonico et al. (2014). Fourth, our findings are likewise robust to changes in the composition of the sample, such as excluding different segments of the line of contact (i.e., land vs river), different states, or municipalities directly adjacent to the line (Barreca et al., 2011). Fifth, placebo estimates based on randomly-drawn lines or plausible alternative lines based on salient geographical features indicate that our results are unlikely to have arisen due to chance.

These findings are remarkable due to the short-lived nature of the treatment. The Allied occupation lasted three months at most and 75 days on average. Naturally, this brief period of occupation corresponds to a bundled treatment, making it difficult to fully adjudicate among competing explanations. However, by combining information from multiple sources – including several pieces of newly-collected data – we can shed light on some of the most plausible mechanisms, which we interpret through the lens of a stylized model of political agency.

To examine potential persistent effects of Allied exposure, we collected new data on more than 1,300 mayors and show that Allied-appointed mayors were swiftly replaced following the Soviet takeover. We corroborate this finding by fielding a retrospective survey among residents of the former GDR to shed light on the policies associated with the temporary Allied occupation. In line with the historical record, a comparison of responses across the line of contact shows that the Allied occupation is more positively perceived and is associated with better governance and higher public good provision. Focusing on the Allied-occupied area, we also observe that a majority of respondents report a deterioration in governance after the Allied withdrawal and an overall negative perception of the Soviet take-over.

We explore and find less conclusive evidence for additional mechanisms. Using data from the 1946 population census in the Soviet occupation zone, we rule out that differential migration and changes in the economic structure across the line of contact are key drivers of our observed effects. Similarly, differences in the exposure to the radio signal originating in the Allied sector (i.e. access to media from *outside* the GDR) do not drive our main results. We also find no persistent impacts on public good provision (as measured using the stock of housing) or on the intensity of political control (as measured by the presence of *Stasi* informants and arrests) during the GDR period. Finally, we use data on the 1946 elections in Berlin to show that the local population did not simply become attached to the initial occupying force (i.e. a "victor effect").

This paper contributes to several strands of the literature. First, we contribute to the growing empirical literature on the determinants of costly political opposition in nondemocracies. A large body of work has studied the effects of access to free media or, more recently, social media (Kern and Hainmueller, 2009; Crabtree et al., 2018b; Enikolopov et al., 2020; Gagliarducci et al., 2020). Other papers have focused on the impact of state repression (Bautista et al., 2020; Rozenas and Zhukov, 2019). A recent strand has also examined the role of social interactions, peer effects and habit formation in protest participation (Cantoni et al., 2019; Stegmann, 2019; Bursztyn et al., 2021). We add to this literature by showing that brief exposure to better governance and a disciplined occupying force has a lasting effect on costly resistance to dictatorship.²

Our results also speak to the literature on military-civilian interactions in conflict and post-conflict settings (Berman and Matanock, 2015). Previous work has shown that heartsand-minds strategies are more effective than overwhelming firepower in winning over the local population (Berman et al., 2011; Kocher et al., 2011; Ferwerda and Miller, 2014; Fontana et al., 2017; Dell and Querubin, 2018). We complement this literature by showing that short-lived differences in the quality of military governance in the very early stages of post-conflict reconstruction can have a lasting effect on local political attitudes. In this regard, our findings also provide evidence on the importance of initial conditions for nation-building.³

Finally, we contribute to understanding the effects of the division of Europe among the

²Our findings are consistent with previous work on the lasting effects of short canvassing interventions on beliefs and political outcomes in established democracies (Broockman and Kalla, 2016; Pons, 2018). ³See e.g. Suelik (2012): Alazing et al. (2021): Barri et al. (2010)

 $^{^{3}}$ See e.g. Svolik (2013); Alesina et al. (2021); Bazzi et al. (2019).

victors of World War II. Previous important work has studied the economic and political consequences of changes in the identity of the occupying force in Austria (Ochsner, 2017; Ochsner and Roesel, 2020), West Germany (Schumann, 2014) and Czechoslovakia (Grossmann et al., 2021).⁴ Similar to these papers, our identification strategy relies on exploiting border discontinuities in a spatial RDD. Although these studies do not directly focus on political resistance to authoritarianism, they suggest that population flows could be an important part of the mechanism underlying our results. The episode that has attracted the most academic attention is arguably Germany's partition after World War II (Fuchs-Schündeln and Masella, 2016; Becker et al., 2020). While most work has relied on East-West comparisons between the (democratic) Federal Republic of Germany and the (autocratic) GDR (e.g., Alesina and Fuchs-Schündeln, 2007; Laudenbach et al., 2020), our empirical strategy provides a new perspective by using the line of contact as a source of variation *within* the GDR.⁵

The remainder of the paper is organized as follows. In section 2, we provide the historical background to situate our results. In section 3 we introduce the data; section 4 discusses the spatial regression discontinuity design and section 5 presents the main results. Section 6 provides a discussion of potential mechanisms and section 7 concludes.

2 Background

2.1 The End of World War II in Germany (Jan-May 1945)

On May 7 1945, Germany surrendered to the *Grand Alliance* led by the United States, the United Kingdom and the Soviet Union, putting an end to World War II (WWII) in Europe.⁶ Preparations for the German surrender had begun months in advance, including discussions on the temporary division of the country's territory into separate occupation zones for each of the victors. The final arrangement was ratified at the Yalta Conference in February 1945,

⁴Ochsner (2017) exploits the line of contact in Austria to find that even brief contact with the Red Army led to a relative population decline in the long-run. Schumann (2014) exploits differences in resettlement policy between the French and American occupation zones to study the persistence of population shocks. Ochsner and Roesel (2020) exploit the Danube river separating US and Soviet forces to show that the outmigration of Nazis to the US occupation zone fostered right-wing party success in the receiving Austrian municipalities. Finally, Grossmann et al. (2021) compare areas occupied by the US and the USSR in a spatial RDD to show persistent impacts of anti-fascist Germans staying in post-war Czechoslovakia on modern-day Communist Party support.

⁵One exception is Eder and Halla (2018), who document the impact of the demarcation line on longrun population dynamics at the county-level. Our analysis, focusing on shorter term political outcomes, is conducted at the granular municipality level which is more suitable for an RDD.

⁶This section draws largely on Ziemke (1976); MacDonald (1993); Ambrose (2000); Beevor (2002); Toland (2003); Henke (2009); Jones (2015).

but initially not made public. It allocated the eastern states of Brandenburg, Mecklenburg-Vorpommern, Saxony-Anhalt, Saxony and Thuringia to the Soviet occupation zone. The German capital, Berlin, was also divided into separate sectors for each victor, despite being located well within the Soviet zone.

Importantly, the decision was made to have the boundaries of the occupation zones only come into effect after the German surrender and to not allow them to interfere with ongoing military operations. As a result, the occupation zones provided little guidance on the final location of the Soviet Red Army and the Allied Expeditionary Force (AEF) when the war came to an end. On the western front, the Allies had been making progress through France and Belgium since the D-Day landing in June 1944. On the eastern front, the Red Army had been on the offensive since mid-1943. In the fall of 1944, rapid Allied advances suggested that the *line of contact* where Soviets and Allies would meet might be as far east as pre-war Poland, but the Soviet offensive later that year plausibly pushed it as far west as the river Rhine. When the Red Army reached the river Oder at the end of January 1945, only 80 km away from Berlin, it seemed certain that it would be able to capture the entire Soviet zone before linking with the Allies (Panel a in Figure 1).

Preparations for the capture of Berlin and bad weather would hold back the final Soviet offensive for over two months. Meanwhile, the Allies made rapid progress through western Germany, especially after capturing a bridge over the Rhine at Remagen (480 km from Berlin) on March 7. In early April, Allied troops entered the Soviet zone in western Thuringia (Panel b in Figure 1). On April 12, they reached the river Elbe near Magdeburg and the river Mulde, one of the Elbe's smaller tributaries, south of Dessau. Allied troops were roughly 100 km away from Berlin, deep inside the Soviet zone, and Allied commanders started making preparations for the final assault on the capital.

The decision whether to capture Berlin fell on Allied commander Dwight Eisenhower. Eisenhower had complete authority over Allied operations and prioritized military criteria over geo-political considerations. He decided that Berlin was excessively costly and on April 15 ordered his troops to halt the advance and allow the Soviets to capture the city.⁷ Fearing the possibility of friendly fire as the Red Army approached, Eisenhower informed the Soviet High Command on April 21 that Allied forces would hold the north-south line formed by the rivers Elbe and Mulde all the way to the Czech border.

The Soviet attack on Berlin had begun on April 16 (Panel c in Figure 1). Over the

⁷Despite pressure from British Prime Minister Winston Churchill to "shake hands with the Russians as far to the east as possible" (Beevor, 2002, p.195), Eisenhower aimed to minimize casualties given the ongoing war against Japan in the Pacific. He was further dissuaded by the fact that any territorial gains would only be temporary. The Soviets' explicit interest in capturing Berlin, mostly driven by the capital's symbolic significance, also played a part.

following days, Soviet troops encircled and fought their way through the city. At the same time the Red Army resumed its westward push towards the Allies. The first contact between Allied and Soviet troops took place at Torgau on April 25, as scattered Allied patrols raced to be the first to find the Red Army (Panel d in Figure 1).⁸ Fearing a Soviet drive into Denmark, the Allies made a last-minute change to the initial plan by crossing the Elbe into the northern state of Mecklenburg-Vorpommern on May 1 (Panel e in Figure 1), where they would occupy positions all the way to the Baltic.⁹ In the final days of the war, German soldiers and civilians made a westward dash in an attempt to avoid capture by the feared Red Army, but the Allies imposed tight controls on movement across the line.¹⁰ By the time of the final German surrender on May 7, the gap between the two armies had been all but closed (Panel f in Figure 1).¹¹ At this time, almost 40% of the Soviet occupation zone was under Allied control (Ziemke, 1976).

2.2 American occupation of the Soviet zone (May-July 1945)

Following the German surrender, the local population remained uncertain about the duration of foreign occupation and the final identity of the occupying force. Details on the partition of Germany only emerged by early June when the Soviet Union pressured the Allies to vacate the designated Soviet zone.¹² Whether the Allies would comply was also unclear, with the British government secretly arguing against leaving before all agreements with the Soviet Union had been reached. Some rumors about an American retreat began spreading among civilians, but even the newly-appointed civilian authorities and local military commanders were very much left in the dark.

Ultimately, US President Harry Truman opted to abide by the Yalta agreement and informed the Soviet leadership on June 14 of his intention to withdraw from the Soviet zone

⁸Those patrols only rushed forward to meet the Red Army, but the Allied forces otherwise remained at the stipulated line, about 25 km west of Torgau along the river Mulde.

⁹The Allies also moved slightly east of the Mulde in a small segment where the river bends west near the city of Chemnitz in the state of Saxony, apparently as a result of confusion among ground troops between two arms of the river, *Zwickauer Mulde* and *Freiberger Mulde*.

¹⁰At Lüneburg Heath, the British refused to accept the partial surrender of three German armies trying to escape the Russians in Mecklenburg. At the Elbe, where many bridges were destroyed and Allied troops tightly controlled crossings, the American commander agreed to allow German soldiers to cross, but not civilians. As the Russians approached and fighting intensified, many people drowned trying to cross the river on improvised rafts or by swimming.

¹¹In Erzebirgskreis, near the Czech border, a small pocket remained accidentally unoccupied by either army until late June 1945 (Naimark, 1997). The area, uncoloured in the map, became later known as the *Free Republic of Schwarzenberg*. As it was initially neither Allied nor Soviet-occupied, we drop this section in the main analysis. In the appendix we show that results are robust to including it and assigning it to be Soviet-occupied, i.e. the identity of its first occupier (see Appendix Table A1).

 $^{^{12}}$ A map of the occupation zones appeared on the front page of *The Times* on June 7, having been published by a Russian newspaper the day before. See Appendix Figure A1 for excerpts of the cover.

in exchange for access to Berlin. Military commanders met on June 29 to finalize details and agreed that the Allied forces would sequentially retreat over a four-day period, with Soviet forces following closely behind. Once this process was complete, the Allies would be allowed into West Berlin. The redeployment took place between July 1 and 5, with little forewarning to the local population (Henke, 2009). For example, British forces in Schwerin announced on June 30 that the city would be handed over to Soviet control the following day. Less than two months after the end of the war, the Soviet Union was in full control of its designated occupation zone.

2.3 The 1953 Uprising

The Soviet Military Administration in Germany (SMAD, by its German acronym) was created in June 1945 to administer the Soviet occupation zone.¹³ After four years of Soviet occupation, the German Democratic Republic (GDR) was born on October 1949. By then, the Soviet-backed Socialist Unity Party (SED) had fully consolidated power and the country was well on track to becoming a one-party state along the lines of the Soviet model.¹⁴ Besides the absence of competitive elections after 1946, the country increasingly resembled the Soviet police state following the creation of the secret police, known as *Stasi*, in 1950 (Lichter et al., 2021b). Reparations to the Soviet Union, including the dismantling of factories and infrastructure, crippled the economy (Becker et al., 2020). Hundreds of thousands of people left for West Germany each year.

In 1952, the SED made explicit its aim to engage in the "accelerated construction of socialism" (Grieder, 2012; Steiner, 2013). Following this announcement, the collectivization of agriculture and the development of heavy industries intensified, as well as the fight against religious organizations. On May 14, 1953, the SED increased work norms (i.e. expected output) by more than ten percent without extra compensation (Kopstein, 1996). Under Soviet pressure, it reversed course on June 11, while at the same time publicly admitting to excesses and mistakes in the official party newspaper *Neues Deutschland*. This admission of weakness made the population believe "that an opportunity existed to affect political change" (Bruce, 2003, p.172). Combined with the fact that the tightened work norms were not modified at first, this triggered a series of protests and strikes throughout the country between June 16-21.

¹³Matters affecting the whole of Germany were decided by the Allied Control Council, with participation of the four victorious powers: US, UK, USSR and France. Berlin was jointly ruled by a *Kommandantura* including representatives of all powers.

¹⁴Other parties were purged of opposition leaders and pressured to join a national coalition, known as the National Front (NF). The NF would present a single list of SED-vetted candidates in all pseudo-elections to take place in the GDR until 1990.

Protests began in East Berlin and quickly spread throughout the country. Protesters attacked SED buildings and freed political prisoners, while demanding the resignation of SED chairman and *de facto* dictator Walter Ulbricht. Industrial workers from hundreds of factories walked out and joined strikes. Estimates of the number of participants in these events range from 0.5 to 1.5 million people, most of them from working class backgrounds (Stibbe, 2006; Grieder, 2012; Steiner, 2013). One common thread linked these different events: "the demonstrators' demands revealed a desire for change in the political system. In the vast majority of disturbances, the demonstrators demanded free elections" (Bruce, 2003, p.182).

After the first day, the government declared a state of emergency and the Soviet military intervened to restore order. According to Grieder (2012), "the fledgling GDR would have collapsed had it not been for the intervention of the Soviet army" (p.40). Bruce (2003) adds that "the revolution in Brandenburg was so extensive that the police candidly admitted thet they thought that the government had fallen" (p.186). Importantly, all other parties had been co-opted by the SED and their leaders driven into exile or underground resistance "leaving nobody of any standing who could negotiate with the SED on behalf of the rebels" (Stibbe, 2006, p.49). More than 6,000 people were arrested and about a dozen were killed during the uprising. Repression intensified in the following months. Arrests doubled and alleged anti-communist saboteurs were seized by Soviet forces and shipped to Siberia. A series of purges within the SED further tightened Ulbricht's hold on power. As a result, the 1953 uprising was the only major episode of political unrest to take place in the GDR until the events that led to the end of the dictatorship and German reunification in 1989/90.

3 Data and treatment assignment

In this section, we introduce our main data sources. We describe other data sources in the relevant sections below. Online Appendix A provides detailed information on all the data used in the paper, including sources, access conditions, and coverage.

3.1 Data sources

Baseline characteristics. To test for the idiosyncratic nature of the line of contact, we consolidate data from existing studies. We rely on Falter and Hänisch (1990) for measures of pre-war population and electoral outcomes. These authors collected socioeconomic data at the county and municipality level for the German Reich between 1920-1933.¹⁵ The underlying

 $^{^{15}\}mathrm{Falter}$ and Hänisch (1990) provides a detailed description of this data.

data is drawn primarily from the statistical office of the Weimar Republic (*Statistisches Reichsamt*) and includes both election reports and census publications. We use data on population size, labor force participation, industry structure, and religious composition from the censuses of 1925 and 1933. We also use electoral data from the last federal elections in 1933.

We complement the pre-war socioeconomic data with WWII measures. First, we use rich information on Allied aerial bombings collected by the Theatre History of Operations Reports (THOR) project from the US Department of Defense (2016). The dataset is constructed from digitized Allied mission reports, providing granular data on the location and timing of aerial bombings between 1939-1945. We aggregate the data and compute whether aerial bombings occurred in proximity to a given municipality. Second, we examine whether synagogues were destroyed in a given municipality during the 1938 November Pogroms using records from the Synagogue Memorial "Beit Ashkenaz". Third, we use the location of all known concentration camps and compute their proximity to each municipality, based on records from the Jewish Virtual Library. Finally, we calculate for each municipality the geodetic distance to a range of relevant locations: Berlin, the inner German border, the closest coast and the nearest mineral reserve (Schulz and Briskey, 2005).

1953 Protest data. The data on the 1953 uprising comes from Crabtree et al. (2018a), which is based on the historical compilation by Kowalczuk (2003). The data contains indicators for the occurrence of protests between June 16-21 in all municipalities with a population above 1,000. Out of 2,585 municipalities considered, 528 experienced a protest. While at a granular level, a shortcoming of the data is that it does not include the size, frequency or exact day of the protest. Our analysis is thus restricted to treating protests as a binary variable and focuses on protest incidence. We also use the Crabtree et al. (2018a) measure of signal strength by the Radio in the American Sector (RIAS) for a robustness check.

3.2 Reconstructing the line of contact

To identify the line separating areas occupied by the western Allied Expeditionary Force and the Soviet Red Army, we consulted multiple historical sources. Based on these sources, we reconstructed the progress made by each army at the most fine-grained level possible (often, but not always, at the city-day level). Despite the abundant literature on the end of WWII in Europe, to the best of our knowledge this information had not been previously harmonized or digitized. Online Appendix A provides a detailed documentation of the establishment of the *line of contact* with a full list of sources. MacDonald (1993) is particularly comprehensive and contains detailed accounts of Allied military advances and maps tracking the front line at various locations and points in time. Also useful were the daily progress reports produced by the Supreme Headquarters Allied Expeditionary Force (SHAEF).

A large part of the line of contact is defined by the rivers Elbe and Mulde. This includes the entire segment through Saxony-Anhalt, as well as part of Saxony. For areas that were not determined by this natural geographic boundary (38% of the line), we consulted historical accounts. These include the segment through Mecklenburg-Vorpommern that originated in the Allied dash to the Baltic and parts of Saxony where Allied officers got confused among the many small rivers and slightly deviated from Eisenhower's plan.¹⁶ Whenever historical accounts did not document the precise course of the line at the municipality level, we consulted municipality-level sources, such as local archives or municipalities' websites containing information on the identity of the initial occupying army. Ultimately, we were able to establish the location of the line at the finest level of 1946 municipalities (N =12,090). Figure 2, Panel a shows the treatment assignment for modern-day municipalities (N = 2,664).

Figure 2, Panel b shows the spatial distribution of protests in 1953 across municipalities with population above 1,000. As the figure shows, the data has robust coverage of protests throughout East Germany. More importantly, the figure displays substantial variation in the incidence of protests across the country. We observe protests in the less populated northern part as well as in more densely populated areas in the south. Furthermore, the raw data provides some preliminary evidence that protests are more likely to be concentrated on the Allied side. The binary nature of the protest measure, however, complicates a conclusive visual inspection. We thus resort to testing for the discontinuity more formally using a standard spatial regression discontinuity design.

4 Empirical strategy

The line of contact between Allied and Soviet forces at the end of WWII provides idiosyncratic variation in the areas exposed to Allied and Soviet occupation. We exploit this variation in the identity of the initial occupying force in a spatial regression discontinuity design (RDD). As discussed in section 2, the variation delineating the exact frontier between the two armies depended on largely idiosyncratic factors that affected the particular direction and speed at which both armies progressed. We verify below that the final line is orthogonal to pre-determined local economic and political conditions.

To estimate the effect of exposure to the Allied (relative to Soviet) occupation on measures

¹⁶As part of our robustness checks, we verify that the results hold when we exclude (i) segments of the line, (ii) entire states, or (iii) municipalities in the immediate vicinity of the line of contact.

of subsequent political resistance, we implement a cross-sectional spatial RDD (Dell, 2010; Dell and Querubin, 2018). We estimate the following regression,

$$y_i = \alpha + \beta \times Allied_i + f(\mathbf{g}_i) + \gamma' x_i + \varepsilon_i \tag{1}$$

where y_i is an outcome of municipality *i*. Allied_i is a dummy that takes the value 1 if the municipality was initially occupied by the Allied forces (to be later ceded to the Soviets) and 0 if the municipality was occupied by the Soviet forces throughout.

Given the geographic pattern of the foreign occupation of Germany, a comparison between all areas occupied by the Allied vs. Soviet forces is likely to be confounded by East-West baseline differences, such as distance to the inner German border, proximity to Berlin or other underlying socioeconomic differences (Becker et al., 2020). To obtain a valid counterfactual, the regression discontinuity design instead compares only areas in the immediate vicinity around the line of contact. Intuitively, we leverage the fact that geographic and socioeconomic characteristics are smooth (and thus comparable) around the cut-off, whereas the line of contact provides a discontinuous change in the identity of the initial occupying force. Comparing an area that happened to be *just within* the reach of the Allies to an area that fell *just short* will thus provide the causal effect of Allied occupation around the line of contact.

To implement the spatial RDD, we first restrict the sample to a narrow 50 km corridor along the line of contact.¹⁷ We then follow standard practice by smoothly controlling for the geographical gradient using the function $f(\mathbf{g}_i)$. In the most basic specification, $f(\mathbf{g}_i)$ controls linearly for the distance to the line of contact, while our preferred specification allows the slope to vary flexibly on either side of the cutoff as is customary. We also verify that our results are robust to using higher order polynomials or a non-parametric estimator for $f(\mathbf{g}_i)$ following Calonico et al. (2014). In the full specification, the vector \mathbf{x}_i includes a more comprehensive set of covariates to control for the "smoothness" of geography: linear controls for latitude and longitude, border segment fixed effects, as well as distance to Berlin, the nearest coast and the inner border. \mathbf{x}_i also includes additional controls for pre-determined socioeconomic characteristics. The full list of controls is described in the following subsection where we test for the smoothness of the baseline covariates. We cluster the standard errors at the county level. While the assignment to treatment is at the (lower) municipality level, our level of clustering allows us to account for idiosyncratic correlation in the error term among municipalities located in the same county. To more conservatively account for spatial autocorrelation, we also present Heteroskedastic and Autocorrelation Consistent (HAC)

 $^{^{17}}$ We verify below that our results are robust to different bandwidths, including the optimal data-driven bandwidth determined using the selection procedure developed by Calonico et al. (2014).

standard errors following Conley (1999).

4.1 Balance around the line of contact

Smoothness of covariates. The main identifying assumption of the spatial RDD is that the "treatment assignment" - the initial occupation by the Allied (as opposed to Soviet) forces - is as good as random around the line of contact (Imbens and Lemieux, 2008). In this section, we test for balance of the treatment assignment on either side of the line of contact. We conduct the analysis at the municipality level whenever the data permits. This is the finest level of granularity for which data is available. Most pre-war socioeconomic indicators, however, are only available for larger municipalities with a population above 2,000 (Hänisch, 1989).

Table 1 reports descriptive statistics for our dataset. In Panel A, the units of analysis are pre-war municipalities with a population above 2,000 as published by the statistical office of the Weimar Republic. In Panel B, the units of analysis are municipalities of 1946, without any size restriction. In column 1, we produce the means for each of the baseline covariates of interest for the full sample of East Germany. Column 2 reports the difference in means between Allied- and Soviet-occupied areas. We observe that the areas occupied by the Allies are significantly different from those occupied by the Soviets, consistent with the presence of an East-West gradient in socio-economic characteristics (Becker et al., 2020). For instance, Allied-occupied areas had a higher share of industrial workers in 1933, a higher KPD vote share in the 1933 elections and were also more likely to have experienced aerial bombings during WWII. Hence, a "naive" comparison between Allied- and Soviet-occupied areas in East Germany is likely to yield selection bias. In column 3, we thus provide a narrower comparison by comparing the characteristics of Allied- vs. Soviet-occupied areas in a narrow 50 km corridor along the line of contact. We also add the flexible forcing variable which controls for the linear distance to the line and allows the gradient to differ on either side of the cut-off, as well as the segment fixed effects (each of these segments is on average 45 km in length). As column 3 shows, municipalities close around the line of contact are much more comparable, with no major discontinuities in baseline characteristics. Put differently, the covariates are smoothly distributed around the cut-off, conditional on the control function. This assures us of the idiosyncratic nature of the line.¹⁸

 $^{^{18}}$ It is also common in RDD settings to test for sorting around the cut-off. In the spatial context, municipalities naturally cannot strategically sort. But bunching in the density of municipalities may still emerge if the final line of contact was strategically drawn to include or exclude certain locations. We can assess the extent of such bunching by inspecting the distribution of municipalities around the line of contact. Appendix Figure A2 tests the smoothness in the density of municipalities using the local-polynomial estimation proposed by Cattaneo et al. (2020). We find no evidence for a discontinuity.

5 Main results

5.1 First stage: Exposure to Allied forces

We first quantify the difference in the duration of exposure to the Allied forces along the line of contact. We make use of detailed data on the arrival dates for the Soviet Red Army and the Allied Expeditionary Force that we collected for larger cities in East Germany.¹⁹ On the Allied side, there is little variation in exposure to Allied forces when comparing, for example, areas 50 km vs. 2 km away from the line of contact.²⁰ When comparing areas *just* on either side of the line of contact, we observe a difference in exposure of about 70 days.

To quantify the magnitude precisely, Table 2 uses the RDD described in section 4. The RDD confirms that there is a discontinuous jump in exposure to the Allies at the line of contact of around 73 days. The magnitude of the estimate varies little if we simply compare means on both sides of the line of contact (column 1) or if we use a linear forcing variable that controls for the distance to the line (column 2). This magnitude remains comparable once we allow for a more flexible and richer set of forcing variables. In column 3, we flexibly fit the forcing variable on each side of the cut-off; in column 4, we divide the line into ten segments and use border segment fixed effects to further constrain the comparison to a narrow geographic area. Column 5 includes the full set of fine-grained baseline controls shown in Panel B of Table 1, plus latitude and longitude. Not surprisingly, given the balance of covariates, the results remain largely unchanged. Column 6 expands the polynomial for the running variable to a quadratic, while column 7 uses instead a non-parametric local polynomial with bias-corrected inference following Calonico et al. (2014). The coefficient remains stable across specifications. We show standard errors clustered at the county (Kreis) level in parentheses and Conley standard errors that account for spatial autocorrelation in brackets.²¹ The estimates are quite precise either way. They show that Allied-occupied municipalities were ruled for little over two months by Allied forces before the Soviet troops took over in early July 1945.

5.2 Costly political opposition: the 1953 Uprising

We now turn our attention to our main outcome of interest - the incidence of protests during the 1953 uprising. As mentioned in section 2, this was the only major nationwide

¹⁹Length of exposure is approximated using the date of arrival to the largest city in the county (*Kreis*) in which the municipality is located.

 $^{^{20}}$ Appendix Figure A3 shows raw means for 2 km bins in the 50 km window around the line of contact.

 $^{^{21}}$ Conley standard errors cannot be calculated for the local polynomial estimation in column 7 due to limitations in the Stata package.

episode of political unrest in the GDR until the peaceful revolution of 1989. We investigate the relationship between Allied occupation in 1945 and the geography of this costly act of political opposition by relating protest patterns to the line of contact.

Figure 3 shows the raw probability of protests occurring in a municipality as a function of the distance to the line of contact in granular 2 km bins. The sample comprises all municipalities around the 50 km window with a population of at least 1,000, due to data availability.²² As a visual aid, we overlay fitted polynomials of different degrees. The polynomials indicate a sharp drop in protest incidence when we cross the line of contact from the Allied to the Soviet side.

Table 3 reports the corresponding RDD regressions. In column 1, we report the average difference in means in protest incidence on the Allied vs. Soviet side of the line of contact for municipalities located in a 50 km window around the line, without any controls. On average, the probability of a protest occurring is 8 percentage points (pp) higher on the Allied side than on the Soviet side. In column 2, we introduce the RDD design by adding the linear distance to the line as a forcing variable. With the forcing function $f(\mathbf{g}_i)$ governing the assignment to Allied or Soviet occupation, the resulting estimate reflects the causal difference at the discontinuity. The RDD estimate shows a 16.3 pp increase in protest incidence on the Allied side.

In the remainder of the columns, we increasingly impose more flexible RDD specifications, in line with Figure 3. In column 3, we allow the slope of the forcing variable to vary on either side of the cut-off; in column 4, we restrict the comparison to municipalities in close proximity by including border segment fixed effects. In column 5 we add the set of fine-grained baseline controls. As expected, given the as-good-as-random assignment of the treatment conditional on distance to the line, the inclusion of these measures does not substantially alter the estimates. Column 6 allows for a more flexible functional form of the forcing variable by including a quadratic function. Finally, column 7 reports estimates from the local polynomial RD procedure (Calonico et al., 2014). The RD coefficient is stable across these more flexible specifications. Overall, the results suggest that Allied exposure caused an increase in the incidence of protests during the 1953 uprising of between 14 and 18 pp. Compared to the mean of the dependent variable (22%), this reflects a sizable increase. As before, we provide standard errors clustered by county in parentheses and Conley standard errors that account for spatial autocorrelation in brackets. Estimates are highly precise in all cases.

 $^{^{22}}$ The full sample corresponds to around 22% of all municipalities in East Germany and covers about 77% of the population. Importantly, the distribution does not vary along the line of contact.

5.3 Placebo lines, rivers and robustness checks

We provide a series of robustness checks to validate our research design.

Placebo lines. First, to ensure that our estimates are unlikely to have arisen by chance, we conduct a randomization inference exercise that compares the actual estimated effect against the distribution of placebo effects from 10,000 arbitrary divisions of East Germany. For this purpose, we draw a random set of straight lines through East Germany and assign either side of the line to be Allied- or Soviet-occupied. Such treatment assignment mimics the data generating process that gave rise to the actual line of contact and introduces the same type of spatial correlation. This provides a more conservative randomization inference than the conventional Fisher (1935)–type of randomization assuming an i.i.d. assignment across units. To further match the share of municipalities under Allied and Soviet occupation based on the *actual* line of contact, we restrict the analysis to counterfactual assignments that leave no more than 60% of municipalities on either side of the line.²³ This also ensures that the estimates are not driven by implausible divisions that only assign a small fraction of municipalities to Allied or Soviet occupation. Appendix Figure A4 displays four illustrative placebo divisions of East Germany.

For each placebo line, we recalculate the distance to the fictitious line of contact and restrict the sample to a 50 km bandwidth, as in our main analysis. We then estimate the placebo Allied occupation effect using the full specification. This is the same specification used for the actual division in Table 3, column 5. Figure 4 plots the distribution of placebo effects, which is predictably centered around zero. Reassuringly, we find that our estimated effect for the actual Allied occupation is highly unlikely to have arisen by chance (p = 0.025), which enhances the credibility of our main result.

River placebos. A large part of the line of contact is delineated by the rivers Elbe and Mulde, which were used by the Allies as salient geographical features to avoid friendly fire from the approaching Red Army. This may raise the concern that the differential protest effect we observe is in fact a "river effect" and not the result of the fleeting exposure to Allied occupation. Travel costs, for example, could change discontinuously around rivers, thus potentially confounding our results. We conduct several robustness checks to ensure that our results are not driven by rivers. First, Panel A in Table 4 shows that our main results are robust when breaking down the sample by whether the line of contact is defined by land or river boundaries. While the disaggregate estimates are invariably noisier relative

 $^{^{23}}$ The actual line of contact results in 59.2% of municipalities in the protest data under Allied occupation. The results, however, are similar when we relax the distributional assumption of the placebo samples; when imposing that no more than 75 or 90% of observations are assigned to either side, the *p*-values are 0.026 and 0.041, respectively.

to the full sample (reported for convenience in columns 1-2) due to the smaller sample sizes, the point estimates are stable across the land and river sub-samples (columns 3-6). Indeed, we cannot statistically reject that the point estimates are the same for these sub-samples.

Second, we go a step further by replicating the RDD analysis for *placebo* lines of contact based on other East German rivers. We identified three plausible rivers for this analysis, shown in Appendix Figure A5. In the first scenario, we assume that the Soviets progressed faster than they actually did, pushing the line of contact west to the Saale river. In the second and third scenarios, we assume that the Allies made more headway than they actually did, pushing the line of contact east until either the south-eastern portion of the Elbe or a combination of the Spree and Havel rivers.²⁴ Similar to the actual line of contact, the resulting placebo lines roughly run on a north-south direction. In line with the actual assignment, we define municipalities west of the placebo lines to be Allied-occupied. As before, we restrict the sample to municipalities 50 kilometers around each line. Panel B of Table 4 reports the results from these placebo analyses. We find no difference in protest incidence along any of the placebo river lines. These placebo results are both statistically insignificant and economically small in magnitude, further suggesting the absence of confounding "river effects."

Additional robustness checks. Finally, we conduct a range of further robustness checks. Our results are unaffected if we individually drop any of the ten border segments (Appendix Figure A6); similarly, the results hold when we exclude individual states (Appendix Figure A7). Our results also hold up to a class of standard RDD robustness checks, such as the use of alternative bandwidths (Appendix Figure A8), including the optimal bandwidth following Calonico et al. (2014), or the exclusion of observations in the immediate vicinity to the cut-off (i.e. *donut RDD*, Appendix Figure A9). The latter addresses potential concerns about measurement error in the exact location of the line or small spillovers across the line (i.e. Allied bridgeheads across the Elbe before Eisenhower's decision to halt the advance).

6 Discussion

The fact that fleeting differences in initial conditions can shape costly political behavior in a newly emerging state is striking. In the previous section, we showed that roughly 70 days of Allied occupation caused an increase in protests against the budding dictatorship propped by the Soviet Union in East Germany. Our reduced-form estimates arguably correspond to

 $^{^{24}}$ The actual line of contact departs from the Elbe at Dessau, shifting south along the Mulde, while counterfactual line 2 assumes that the line followed the Elbe in its southeast trajectory.

a bundle of different "treatments" that distinguish municipalities on opposite sides of the line of contact. While the bundled nature of the experience of Allied occupation makes it difficult to fully disentangle the mechanisms, we now explore several potential channels that may be driving the protest results and provide historical and statistical evidence on their importance.

To guide the interpretation of the results, we introduce a simple conceptual framework.²⁵ Consider an agent with characteristics \mathbf{x} who faces a choice over two political regimes. For concreteness, let i = s denote the Soviet regime, while i = a denotes the Allied regime. Furthermore, let subscript t = 0 denote the initial occupation, and t = 1 the final occupation. On the Allied side of the line of contact, we thus have a_0 and s_1 , while Soviet-occupied areas experience no change in the occupier's identity (i.e. s_0 and s_1). Agents with characteristics \mathbf{x} derive utility U(i|x) from the governance provided by regime i. An agent will thus only support the Soviet regime if the benefits from doing so exceed the benefits of the Allied regime, i.e. $U(s_1|\mathbf{x}) \geq U(a_1|\mathbf{x})$.

Our main results, which robustly document a greater protest incidence on the Allied side of the line of contact, suggest that *past* exposure to Allied rule reduces *current* support for the Soviet regime:

$$U(s_1|\mathbf{x}, a_0) < U(s_1|\mathbf{x}, s_0) \tag{2}$$

where $U(s_1|\mathbf{x}, a_0)$ denotes the utility derived by an agent with characteristics \mathbf{x} from Soviet governance, conditional on having been governed by the Allies in the past.

In theory, there are multiple mechanisms through which the inequality in (2) may hold. The brief exposure to the Allies could have, for example, induced the local population to revise downwards the utility derived from Soviet occupation, perhaps because exposure to better governance allowed for more accurate beliefs about the counterfactual. In this regard, the historical record suggests the presence of uncertainty among the local population about what the occupation would entail: "The Americans are coming!... excitement fluctuating between fear and joy" (Gebhardt, 2017, p.32). Alternatively, fleeting exposure to Allied rule might have induced a more behavioral response (Alesina and Passarelli, 2019). For instance, exposure to better governance under the Allies could have modified the reference point used to evaluate Soviet rule or triggered a "near miss" effect (Kahneman and Tversky, 1979; Kahneman and Miller, 1986).

We use this framework to characterize several potential mechanisms. First, we examine if there were any tangible short-run differences in the occupation experience across the line

²⁵This simple model falls under the general class of models of political agency (Ashworth, 2012).

of contact. Second, we consider possible changes in the characteristics of the population around the line of contact resulting from differential migration. Third, we examine whether there were any persistent differences in policies across the line of contact during the GDR period. Finally, we consider a possible "victor effect" by which the local population rewarded whichever incoming army arrived first to their locality.

6.1 Short-Run Differences in Governance

6.1.1 Appointment of Mayors

Occupying forces on both sides appointed municipal mayors shortly after their arrival as a way of promoting a swift return to local self-government. While the SMAD made appointments mostly based on partisanship (Naimark, 1997), the Allied administration may have used selection criteria conducive to higher-quality appointments. If these initial appointments remained in office beyond the Soviet take-over, they could have had long-lasting consequences (Martinez-Bravo et al., 2017).

To investigate differences in the appointment of these local officials, we hand-collected data on almost 1,300 mayors of 173 cities for the period 1934-1955. For this purpose, we consulted the contemporary websites and the historical registers of these cities, as well as biographical encyclopedias (Müller-Enbergs et al., 2010). Whenever data was unavailable from public sources, we contacted the local archive or press office of the city to collect the information. For each mayor, we gathered data on term dates, party affiliation and the authority that made the appointment. Our sample covers 81% of cities with current population above 10,000 inhabitants, with no evidence of selective non-response between Allied- and Soviet-occupied areas (p=0.12).

Consistent with the historical narrative, our data shows stark differences in the partisan affiliation of initial mayoral appointments across the line of contact. On the Soviet side, 53% of the appointed mayors were members of the German Communist Party (KPD) and a further 28% belonged to the Social Democratic Party (SPD). Only 8% of Soviet-appointed mayors had no party affiliation. On the Allied side, only 12% of appointed mayors belonged to the KPD, 32% came from the SPD and 32% had no party affiliation.²⁶ The Allies mostly appointed previous office-holders from the Weimar Republic, who espoused liberal views and had been politically persecuted during the Nazi regime (Schneider, 1989; Welsh, 2010). These differences in the characteristics of mayors constitute preliminary evidence of changes in governance across the line of contact.

 $^{^{26} {\}rm The}$ Allies appointed a mayor in 85% of occupied cities, but this does not predict protest incidence in 1953 (p=0.807).

For the most part, Allied-appointed mayors did not remain in office after the Soviet takeover, with 82% removed before the new year.²⁷ As further evidence of the connection between these dismissals and the Soviet take-over, Panel a in Figure 5 shows the distribution of monthly mayoral appointments in 1945 on both sides of the line of contact. The distribution is bimodal on the Allied side, peaking both in April (when the Allies arrive) and in July (when the Soviets take over). In contrast, the Soviet side has a unimodal distribution peaking in May when the Nazis surrender. Importantly, the new mayors appointed by the Soviets in the Allied area look comparable in their party affiliation to the initial appointments on the Soviet side: 67% of the replacement mayors were from the KPD, and a further 20% from the SPD. This suggests that little remained of the personnel decisions made by the Allies after their withdrawal.

The longitudinal nature of the data allows us to study broader differences in patterns of mayoral turnover across Allied- and Soviet-occupied cities over the two decades between 1934-1953. By looking at the period 1934-1944, we can provide further evidence of balance in pre-determined covariates before 1945. In addition, the data for the period 1946-1953 allows us to examine potential differences in appointments across the line of contact in the lead-up to the 1953 uprising.²⁸ Panel b in Figure 5 shows results from a panel regression using *excess turnover* as the dependent variable, defined as having more than one mayoral appointment on the same year.²⁹ The results show that cities on the Allied side exhibit a 17 percentage point (pp) relative increase in the probability of excess turnover in 1945, which is the only year with an economically and statistically significant effect. This suggests that the pattern documented in Panel a is capturing circumstances unique to the change in occupying force in 1945 and does not reflect deeper underlying differences in mayoral turnover across the line of contact.

6.1.2 Retrospective Survey on Occupation Experience

The previous results confirm that Allied occupation entailed differences in governance (as embodied in local leaders) relative to Soviet occupation across the line of contact. However, data limitations prevent us from directly measuring differences in the *quality* of governance.

 $^{^{27}}$ Appendix Figure A10 shows the final year in office for Allied-appointed mayors, disaggregated by party affiliation. Only 40% of mayors belonging to the left-wing parties (KPD and SPD) were dismissed in 1945, while 95% of mayors with other affiliations and non-partisans were dismissed on this year.

²⁸Unfortunately, the number of cities for which we have information on mayors is too small to credibly implement a regression discontinuity design. However, the longitudinal nature of the data allows us to estimate panel regressions in which we include city fixed effects and year fixed effects to account for persistent cross-sectional differences and common temporal shocks.

²⁹Excess turnover has been shown to reflect political interference in other settings (Iyer and Mani, 2012; Akhtari et al., 2020). Results are comparable when using total yearly appointments per city.

If Allied occupation exposed the local population to better governance, the exposure – even if fleeting – could have fostered opposition to Soviet rule.

To further explore differences in the quality of governance, we collected retrospective data by fielding a survey in early 2020 using a professional survey company. We targeted individuals who either have first- or second-hand experience from the end of WWII by sampling respondents aged 60 or above in East Germany. Overall, we were able to collect responses from 472 individuals.³⁰ Details of the data collection and measurement issues are discussed in Appendix C. While the survey data allows us to complement the municipality- and citylevel analysis with individual data, long-term retrospective surveys come with limitations, the main ones being recall bias and small sample size due to attrition. We thus have to rely mostly on second-hand accounts from children with parents residing in East Germany in May 1945, and are not able to conduct an RDD. However, to enhance comparability and mimic an RDD, we restrict the sample to respondents in a 100km window around the line of contact.³¹

The results are summarized in Figure 6. Panel a reports the mean difference in outcomes between the areas initially occupied by Allied and Soviet forces. Respondents from Allied areas report systematically better perceptions of the occupation experience. In particular, respondents recall higher public good provision by the Allies (food and medicine), though there is no significant difference in the mediation of conflicts. The Allies are also recalled as significantly less likely to engage in misconduct.³²

While the Allied-occupied areas benefited from initial exposure to better governance, these benefits were short-lived. In Panel b of Figure 6 we restrict the sample to respondents in the Allied area and examine their responses to questions about changes in the quality of governance following the Allied withdrawal. We find that the Soviet takeover of the Allied area is accompanied by a large downward revision of the occupation experience. The vast majority of respondents (69%) perceive the switch from Allied to Soviet control as negative (12% perceived it as positive, 19% as neutral). When asked whether the arrival of the new

 $^{^{30}}$ See Appendix Figure A11 for imposed restrictions leading to the final survey sample. Appendix Table A2 compares the survey sample with the East German population with ages 60+. Our sample is positively selected on education and includes slightly more men than women, but it reproduces quite accurately the distribution of the population across states.

³¹The results hold when we use the full sample and are robust to the inclusion of individual-level controls for first/second-hand account, age, gender and education (Appendix Table A3).

³²Misbehavior by Soviet troops in Germany has been documented extensively, particularly sexual violence against women (Naimark, 1997; Beevor, 2002; Gebhardt, 2017). The German army also engaged in sexual violence and other atrocities during its occupation of the Soviet Union (Wood, 2006; Henry, 2011). The effect of exposure to misbehavior on the propensity to protest is theoretically ambiguous. More abuse may lead to more animosity against the perpetrator, but it could also lead to fear of reprisals. Previous work shows that civilians exposed to abuse under dictatorship engage in stronger opposition when a window of opportunity arises (Bautista et al., 2020; Rozenas and Zhukov, 2019).

occupier led to changes in public good provision, we find that the majority of respondents indicate that the new occupying army distributed less food and medicine and 81% report an increase in misbehavior by the occupying force.

These reported changes in the quality of governance are in line with the historical record: "Unlike the... American military government, the [Soviet] kommandanturas often had no specially trained staff and Soviet officers were assigned to regions and jobs about which they had little or no knowledge. There was no evidence of handbooks, and Russian maps tended to be cutouts of outdated German originals" (Naimark, 1997, p.13). Moreover, the survey responses and the historical accounts coincide in awarding a prominent role to misconduct by Soviet forces after the Allied withdrawal: "After the arrival of the Russians... 'Rapes, assaults, murders, one after the other'" (Naimark, 1997, p.85).³³

6.2 Selective migration

Differences in the characteristics of the civilian population driven by selective migration across the line of contact could also help explain our findings (Ochsner, 2017; Eder and Halla, 2018; Ochsner and Roesel, 2020; Grossmann et al., 2021). Table 1 showed that the two sides of the line were highly comparable in their demographic, geographic and political characteristics in the 1920s and 30s, as well as in the impact of the Nazi regime and WWII. However, differential migration to the Allied-occupied area in the immediate aftermath of the war may have led to a disproportionate concentration of opponents to the Soviet regime on that side of the line. In terms of our model, this explanation corresponds to a case in which people with more positive priors over the Allied regime's quality self-selected into the Allied side (i.e. $U(a_1|\mathbf{x}_a) > U(a_1|\mathbf{x}_s)$).

Years of exposure to Nazi propaganda and rumors of atrocities committed by the approaching Red Army generated a strong preference for Allied occupation as the war came to an end in Germany (Naimark, 1997). Beevor (2002) claims that in Berlin "optimists were learning English and pessimists learning Russian" (p.190). Hence, it is certainly plausible that the Allied-occupied area within East Germany saw a substantial inflow of potential opponents of the Soviet regime. These flows had two possible sources. First, a large wave

³³While our survey documents lower initial misbehavior in the Allied areas and an increase in misbehavior after the Soviet take-over, it is unclear whether *total* exposure differed across the line of contact. It is plausible that misbehavior was worst in the first months of the occupation, when oversight over soldiers was weaker, in which case the total exposure to misbehavior would be lower on the Allied side. But it could well be that the Soviet army acted more harshly in the area temporarily occupied by the Allies (see Dehdari and Gehring, 2019, for the case of Alsace-Lorraine), in which case the net effect is ambiguous. The historical record does suggest that "even in those areas initially occupied by the American and British troops... rape became a severe problem for the German population." (Naimark, 1997, p.84). Given data limitations, we are unable to empirically distinguish between these two cases.

of migrants that fled westward during the Soviet invasion of Germany in the final months of the war, mostly from the eastern provinces of Silesia, Pomerania and Prussia. Second, more selective cases of migration to the Allied sector during the initial occupation after the end of the war.³⁴

In this context, a migration-based explanation faces several challenges. Migration from the eastern provinces began during the Soviet winter offensive in late 1944 and peaked in early 1945 (Beevor, 2002). At the time, the Allies were hundreds of kilometers away, the division of Germany into occupation zones remained confidential and the location of the line of contact was highly uncertain. Hence, war-induced migration across the line of contact within East Germany could only be quantitatively meaningful in the final days of the war when the Allies were awaiting the Soviets. At that time, however, transportation was completely disrupted and Allied troops restricted civilian crossings (MacDonald, 1993). More importantly, between the end of the war and the construction of the Berlin Wall in 1961, anyone wanting to escape from Soviet rule could easily do so by crossing into West Berlin (Becker et al., 2020).³⁵

Empirically, we can examine the evidence in support of a migration-based mechanism using data from the population census carried out in the Soviet zone in October 1946 (Statistisches Zentralamt, 1948; Falter, 1997). Aggregate statistics show that 74% of people lived in the same state as in 1939. Refugees from the eastern provinces represented 13% of the total population and a further 8% were foreign nationals (mostly resettled Germans).³⁶

Importantly for our RDD estimates, selective migration can only drive our results if it occurs around the line of contact. Unfortunately, data on migration is only available at the county-level. As Appendix Table A5 shows, the share of people residing in the same state as in 1939 does not vary discontinuously around the line of contact at this level. We can also corroborate the lack of evidence for selective migration using municipality-level information on the composition of the population in 1946 (Table 5). In columns 1-6, each column shows RDD estimates using a different variable in the census as dependent variable. All columns use the same fully specified regression discontinuity model (i.e. Table 3, column

³⁴There is no reason to expect the number of resettled Germans arriving from several countries in Eastern Europe after the end of the war to jump discontinuously at the line of contact, since the expulsion of Germans was triggered by the agreement on post-war Europe reached at the Potsdam conference in August 1945, after the Allied withdrawal.

³⁵According to records of the West German Federal Ministry of Displaced Persons, Refugees and War Victims, more than 2.7 million people relocated from East to West Germany between 1949 and 1961, corresponding to 15% of the 1949 population (Rühle and Holzweißig, 1988).

³⁶Appendix Table A4 shows the decomposition of the population in each state in 1946 by place of residence at the start of the war in 1939. The shares above are fairly comparable across states, except for Mecklenburg-Vorpommern, a largely rural state that experienced a large land reform in 1945 and, as a result, attracted a larger share of migrants (Naimark, 1997).

5). Column 1 shows that the total population, while larger on the Allied side of the line, does not jump discontinuously at the line of contact. Furthermore, the estimate is very similar to the baseline estimate for 1933 population in Table 1, providing evidence of no excess migration to the Allied side of the line of contact. Column 2 shows that the female share of the population is almost identical at the discontinuity. This further suggests that the impact of the war was homogeneous on both sides of the line (i.e. military casualties and prisoners of war). Column 3 shows no evidence of a discontinuity in the share of protestants. This is important, as the eastern provinces of Silesia and Pomerania from where many refugees originated had relatively high shares of Catholics. Columns 4-6 study the sectoral composition of the economy. We do observe ex-post differences in the industry structure: municipalities on the Allied side exhibit lower agricultural employment (column 4), matched by a larger share of workers in services and trade (column 6).

In columns 7-8 we examine the explanatory power of the population characteristics from the 1946 census on the incidence of protests in 1953 using mediation analysis. To facilitate interpretation, column 7 reports the coefficient of Table 3 as a baseline. Column 8 includes all the municipality-level demographic controls from columns 1-6. While these variables themselves may have been impacted by the different occupying armies (and are thus potentially "bad" controls), examining the stability of the Allied coefficient allows us to assess the extent to which variation in the observed unbalanced dimensions can explain away our main effect of interest. Reassuringly, we find that the point estimate is only slightly reduced after introducing those controls. This difference, furthermore, is not statistically significant. While differential migration across the line of contact may have certainly occurred, differences in population size and in the characteristics of the local population cannot explain away our core findings on political resistance.

6.3 Persistence in policies, resistance and media

Another explanation for the stronger opposition to dictatorship in the area initially occupied by the Allies is that the occupation – despite its short duration – allowed for the implementation of policies causing long-lasting benefits to the local population. While theoretically possible, the historical reading suggests that this is unlikely. Following the Nazi surrender, the occupying forces had to attend to a host of pressing issues, including the relocation of displaced persons, the handling of prisoners of war, and the ongoing war in the Pacific. The welfare of the local population was not a priority, as a declassified memo on the US postsurrender program for Germany bluntly acknowledges: "The sole purpose of the military in control of the German economy shall be to facilitate military operations and military occupation" (NARA, 1944). Uncertainty about the duration of Allied presence in the Soviet zone further limited long-run planning and policy implementation (Ziemke, 1976; Henke, 2009).

Differences in Soviet policies. Potential differences in Soviet policies across the line of contact could also potentially explain stronger political opposition on the side initially occupied by the Allies. One prominent policy involved the dismantling of machinery, factories and infrastructure to be shipped back to the Soviet Union as war reparations (Ochsner, 2017). Another prominent policy was a land reform implemented in late 1945, which confiscated large estates without compensation.³⁷ These policies could explain our findings if they disproportionately affected the Allied-occupied sector. Arguably, the discontinuous jump in the agricultural share of workers reported in Table 5 could be the result of the differential impact of these policies. However, the previous results showing that our findings are robust to the inclusion of these characteristics as additional controls indicates that this is not the main mechanism. More generally, the dismantling of industrial facilities and the land reform took place throughout East Germany. The historical record at no point mentions the line of contact or the Allied-occupied sector when discussing the motivations or the implementation of any Soviet policy during the occupation years.

Outcome measures during the GDR. We use outcome measures from the GDR period to test whether the brief period of Allied exposure had any persistent impacts on policies in the longer term. Specifically, we digitized data from a housing census collected in 1971 (Statistisches Bundesamt, 1994). The census lists the stock of houses built in different periods, allowing us to test whether areas briefly occupied by the Allies see differential housing construction. We also leverage data from Lichter et al. (2021a) which provide measures of the spying intensity (as measured by the number of unofficial Stasi informants per capita) and political arrests per capita in the 1980s. Given the general paucity of data from the GDR period, the newly collected data come with limitations: first, the policy outcomes are measured *after* the protests in 1953. As such, any differences found will either reflect highly persistent policy differences or potential policy responses resulting from the protests. Second, the outcomes are only available at the county-level, thus reducing the statistical power of our RD design. Reassuringly, we are able to replicate our main protest result at the county-level, thus validating the coarser county-level RD design.

The results are reported in Table 6. We run the same spatial RDD regression as in the municipality-level specification (see Equation 1), restricting the counties to the sub-sample that lies within 50km of the line of contact. The outcome in column 1 is the share of municipalities in a given county that report a protest in 1953. Confirming our main finding,

 $^{^{37}}$ Naimark (1997) estimates that 25% of industrial facilities were dismantled during the summer and fall of 1945. The land reform targeted the *Junker* elite thought to be complicit with the Nazi regime.

we detect the same positive and significant protest effect: counties on the Allied side of the line of contact exhibit greater subsequent regime resistance in 1953. In columns 2-3, the dependent variable is the (log) number of housing units available in the county. Column 2 examines housing built before 1946, while column 3 focuses on housing built on or after that year. In 1971, there was no statistically significant difference between Allied vs. Soviet occupied areas in the stock of housing built before 1946 (column 2). This result is consistent with the interpretation that the fleeting Allied occupation was too short-lived to have led to substantial public works, in line with the historical narrative.³⁸ In column 3, the dependent variable is the (log) number of housing units constructed in 1946 or after.³⁹ Again, the results do not show a significant difference in public good provision, as measured by housing construction.

In columns 4-5, we look at spying intensity and political arrests in the final decade of existence of the GDR as additional policy measures. Interestingly, we find no statistically significant differences in the deployment of Stasi informants (column 4); similarly, we do not find that areas that were under the fleeting administration of the Allies saw a persistently larger number of political arrests (column 5). While not conclusive given the limitations described, these combined results provide no evidence for persistent and large differences in policies after the short-lived period of Allied occupation.⁴⁰

Differences in resistance and access to free media. Even if Allied-appointed mayors were quickly replaced after the Soviet take-over, Allied occupation may have fostered subsequent resistance to the dictatorship if it allowed for the development of a robust network of pro-democracy groups or clandestine resistance cells. These groups could have facilitated co-ordination and enabled civil society to overcome free-riding problems in protest participation (Olson, 1965; Tullock, 1971; Tarrow, 1998).

While also theoretically plausible, this mechanism does not find much support in the historical record. First, the short duration of the Allied occupation combined with the uncertainty about the date of withdrawal arguably hindered the Allies' ability to develop strong resistance groups. As mentioned above, the Allied leadership had other short-term

³⁸The stock of housing in 1971 that was built before 1946 potentially conflates differences in the stock of housing before WWII, differential wartime destruction, and differences in early post-war reconstruction. However, Table 1 suggests balance in socio-economic characteristics before the war and comparable exposure to wartime bombing, thus providing support for our interpretation.

³⁹Unfortunately, the data does not allow us to distinguish between the period before 1953 and the period after that year, which would have enabled us to more precisely study a strategic response by the regime to the greater incidence of protests on the Allied side during the 1953 uprising.

⁴⁰In Appendix Table A6, we also relate the historical exposure to Allied occupation along the line of contact to protests in 1989 and voting outcomes in 1990. While the point estimates are consistent with higher protest incidence in 1989 and lower support for the successor party of the SED (PDS) in 1990, the results are imprecisely estimated and statistically insignificant.

priorities and only slowly awakened to the new geopolitical reality (Ziemke, 1976; Ambrose, 2000). Second, the inner German border remained open and fairly unregulated at least until 1952 (the border between East and West Berlin until 1961), suggesting that American intervention in East German politics was unlikely to be determined by the fleeting line of contact in 1945 (Murphy et al., 1997). Third, while there certainly were open and clandestine attempts at organized resistance to the budding dictatorship in East Germany, the Soviet interior ministry (NKVD) was highly effective at neutralizing any attempt at organized opposition (Bruce, 2003; Gregory, 2009). The 1953 uprising posed such a large threat to the regime precisely because of its spontaneous and uncoordinated nature. Lastly, there is no evidence of Allied backing or intervention in the 1953 uprising: "Western agents... were not responsible for causing the unrest in the first place... the East German rebels of 1953 neither requested nor expected military assistance from Washington" (Grieder, 2012, p.42)

A related topic that has attracted substantial academic attention is exposure to free media originating outside of East Germany, including West Berlin (Kern and Hainmueller, 2009; Bursztyn and Cantoni, 2016; Crabtree et al., 2018b). To the extent that the availability of Western media varied discontinuously around the line of contact (e.g. through the strategic placement of signal towers by the Allies), our results could reflect greater exposure to Western media. To investigate this possibility, we use the measures of exposure to the *Radio in the American Sector (RIAS)* available from Crabtree et al. (2018a). The results in Appendix Table A7 show that RIAS signal strength decreases with distance to Berlin, but does not vary discontinuously at the line of contact. More importantly, our main result is unchanged if we control for the RIAS signal strength.⁴¹

6.4 The "victor effect" – variation within Berlin

One explanation for the increased opposition to the SED dictatorship in the Allied-occupied area is that residents of this area were grateful to Allied rather than Soviet forces for dislodging the Nazis. Internal opponents of the regime, who had faced strong repression in previous years, may have developed sympathy to whichever foreign army they saw first. Moreover, even though Nazi rule enjoyed widespread popular support for more than a decade, civilians may have been grateful for the end of the long and costly war (Stargardt, 2015). If this is the case, initial Allied occupation would mechanically tilt the balance of popular support against the Soviets (i.e. $U(a_1|\mathbf{x}, s_0) < U(s_1|\mathbf{x}, s_0) < U(a_1|\mathbf{x}, a_0)$). This hypothesis also implies that we should see no difference in support for the Allied regime between Soviet-controlled East

⁴¹Allied occupation may have plausibly increased the *demand* for Western media. While we do not find evidence of a complementary effect on protests between Allied occupation and RIAS signal strength, data limitations prevent us from directly testing for differential media demand.

Berlin and Allied-controlled West Berlin, as the entire city was initially occupied by the Soviet Red Army.

To test this hypothesis, we focus on the Berlin state election held on October 20, 1946. This would be the last citywide election to take place before German reunification.⁴² A distinguishing feature was that Berlin was jointly ruled by a council including representatives of both Western Allies and Soviets. As a result, the forced merger of the German Communist Party (KPD) with the more popular Social Democratic Party (SPD) imposed by SMAD in the Soviet zone to create the SED did not apply within the city.⁴³ The SPD won the election with a resounding 48.7% of the votes, while the SED (i.e., rebranded KPD) came in third with 19.8% of the votes, behind liberal party CDU.

For this part of the analysis, we collected electoral statistics for Berlin from the Department for Statistics Berlin-Brandemburg to construct a balanced panel at the district-election level. The dataset covers 75 districts over four elections: 1929 municipal elections, 1930 and 1932 federal elections, plus the 1946 state election. Forty of these districts are located in the western sector ceded to Allied rule in July of 1945 (i.e., West Berlin). To study the effect of initial Soviet occupation on electoral support for the SED in 1946, we implement a difference-in-differences (DiD) design comparing the difference in SED vote share between East and West Berlin, before and after WWII. We use the KPD vote share in the pre-war elections as the relevant comparable measure to the SED vote share in 1946. Crucial to this comparison is the fact that voters in all Berlin districts were able to vote for the same set of parties at each point in time. Our DiD specification includes district and year (election) fixed effects.

Columns 1-3 in Table 7 show results of a simple comparison of means between West Berlin (i.e., Allied) and East Berlin (i.e., Soviet) in 1946. Column 1 includes no controls and shows an 18 pp difference in the SED vote share. In column 2, which controls for latitude and longitude to capture potential geographic gradients in political preferences, we precisely estimate that the SED vote share is 13.6 pp lower in West Berlin. Further restricting the sample to the common longitude support has a negligible impact (column 3). Column 4 provides the DiD estimates. We find that the SED vote share differentially drops by 10.4 pp in West Berlin in 1946. This estimate is also very precise and it is statistically significant at the 1% level. Column 5 provides disaggregate estimates for all elections, leaving 1929 as

 $^{^{42}}$ In Appendix Table A8, we also provide evidence from the elections held in all East German states in 1946. At that time, the permanent separation of the Soviet zone and the consolidation of an authoritarian regime remained uncertain. Still, our suggestive results show that areas on the Allied side of the line of contact saw lower support for the Soviet-backed SED in the 1946 state elections.

⁴³The joint *kommandantura* demanded a referendum for the merger of the KPD and SPD to be valid in Berlin. After 82% of SPD members voted against this, the party was allowed to field its own candidates under the SPD banner in the city election (Braun, 1993; Stivers and Carter, 2017).

the omitted category. Figure 7 provides a visualization of these results. The difference in the KPD (SED) vote share between West and East Berlin remained fairly constant in 1930 and 1932, relative to 1929 (omitted election year). This lends credibility to the identifying assumption of parallel trends underlying the DiD research design. In 1946, however, we observe a large 10.5 pp drop in the SED vote share in Allied-controlled West Berlin, almost identical in magnitude to the baseline estimate in column 4. This pattern is robust and also holds when we again restrict the sample to districts with a common longitudinal support (column 6).

In sum, the evidence for Berlin shows that support for the SED was in fact lower in West Berlin than in East Berlin in 1946, despite Berlin having been captured in its entirety by the Red Army. This result is inconsistent with a "victor effect" leading East Germans to reward whichever army occupied their territory first.

7 Conclusion

George Orwell, in his famous novel 1984, concluded that "so long as [the masses] are not permitted to have standards of comparison, they never even become aware that they are oppressed."⁴⁴ Indeed, in many non-democracies, the lack of an alternative makes the comparison all too hard. One-party states and absolute monarchies leave no formal alternative. Hybrid regimes often suppress the opposition. Limited exposure to well-defined political alternatives can substantially shift the cost-benefit calculus of citizens when deciding to support or resist a dictatorial regime.

This paper goes to the heart of the Cold War to study political opposition to the budding dictatorship propped by the Soviet Union in East Germany after WWII. By leveraging the idiosyncratic line of contact that partitioned what would become the German Democratic Republic into Allied-exposed and non-exposed areas, we show that a "glimpse of freedom" of 70 days can substantially alter the propensity of citizens to engage in costly resistance against an authoritarian regime. Our main result shows that exposure to the Allies increased the likelihood of protests during the 1953 uprising that almost toppled the dictatorship in East Germany.

While shedding light on a unique episode in history, data limitations prevent us from identifying the exact micro-level mechanism through which the exposure to the Allied forces translates into political behavior. By combining newly-collected data on mayoral appointments and a retrospective survey on the occupation experience, we show that the identity of the occupying force affected the characteristics of local leaders and the quality of gover-

⁴⁴George Orwell, 1984, Chapter IX, p. 237. Penguin Modern Classics.

nance. However, we cannot distinguish between a pure informational (learning) mechanism and more behavioral mechanisms. Future work could investigate such mechanisms, for example, by collecting survey data in a contemporary setting, or running more controlled experiments in the lab or in the field.

An intriguing aspect of our findings is that the line of contact predates the formation of the new state - the German Democratic Republic - in which we study subsequent political behavior. As such, our work provides a prologue to the newly emerging and exciting literature interested in the determinants of political behavior in the German Democratic Republic (Stegmann, 2019; Lichter et al., 2021b; Mohr, 2021). The fact that initial conditions matter also has broader implications for modern state building, particularly for newly-created and fragile regimes in post-conflict settings.

References

- Acemoglu, D. and Robinson, J. (2005). Economic Origins of Dictatorship and Democracy. Cambridge University Press.
- Akhtari, M., Moreira, D., and Trucco, L. (2020). Political Turnover, Bureaucratic Turnover, and the Quality of Public Services. Forthcoming at American Economic Review.
- Alesina, A. and Fuchs-Schündeln, N. (2007). Goodbye Lenin (or Not?): The Effect of Communism on People's Preferences. American Economic Review, 97(4):1507–1528.
- Alesina, A., Giuliano, P., and Reich, B. (2021). Nation-building and Education. The Economic Journal, 131(638):2273–2303.
- Alesina, A. and Passarelli, F. (2019). Loss Aversion in Politics. American Journal of Political Science, 63(4):936–947.
- Ambrose, S. (2000). Eisenhower and Berlin, 1945: The Decision to Halt at the Elbe. Norton, New York, NY.
- Ashworth, S. (2012). Electoral Accountability: Recent Theoretical and Empirical Work. Annual Review of Political Science, 15(1):183–201.
- Badsey, S., editor (2000). The Hutchinson Atlas of World War II Battle Plans: Before and After. Helicon Publishing, Oxford, UK.
- Barreca, A. I., Guldi, M., Lindo, J. M., and Waddell, G. R. (2011). Saving Babies? Revisiting the Effect of Very Low Birth Weight Classification. *Quarterly Journal of Economics*, 126(4):2117– 2123.
- Bautista, M. A., González, F., Martínez, L. R., Muñoz, P., and Prem, M. (2020). The Geography of Repression and Opposition to Autocracy. Forthcoming at American Journal of Political Science.
- Bazzi, S., Gaduh, A., Rothenberg, A. D., and Wong, M. (2019). Unity in Diversity? How Intergroup Contact Can Foster Nation Building. *American Economic Review*, 109(11):3978–4025.

- Becker, S. O., Mergele, L., and Woessmann, L. (2020). The Separation and Reunification of Germany: Rethinking a Natural Experiment Interpretation of the Enduring Effects of Communism. *Journal of Economic Perspectives*, 34(2):143–71.
- Beevor, A. (2002). The Fall of Berlin, 1945. Viking Penguin, London, UK.
- Berman, E. and Matanock, A. M. (2015). The Empiricists' Insurgency. Annual Review of Political Science, 18(1):443–464.
- Berman, E., Shapiro, J. N., and Felter, J. H. (2011). Can Hearts and Minds Be Bought? The Economics of Counterinsurgency in Iraq. *Journal of Political Economy*, 119(4):766–819.
- Braun, G. (1993). SBZ-Handbuch: Staatliche Verwaltungen, Parteien, gesellschaftliche Organisationen und ihre Führungskräfte in der Sowjetischen Besatzungszone Deutschlands 1945-1949. De Gruyter Oldenbourg.
- Broockman, D. and Kalla, J. (2016). Durably Reducing Transphobia: A Field Experiment on Door-to-Door Canvassing. Science, 352(6282):220–224.
- Bruce, G. (2003). Resistance with the People: Repression and Resistance in Eastern Germany, 1945-1955. Rowman and Littlefield.
- Bursztyn, L. and Cantoni, D. (2016). A Tear in the Iron Curtain: The Impact of Western Television on Consumption Behavior. *Review of Economics and Statistics*, 98(1):25–41.
- Bursztyn, L., Cantoni, D., Yang, D. Y., Yuchtman, N., and Zhang, Y. J. (2021). Persistent Political Engagement: Social Interactions and the Dynamics of Protest Movements. *American Economic Review: Insights*, 3(2):233–50.
- Calonico, S., Cattaneo, M. D., and Titiunik, R. (2014). Robust Nonparametric Confidence Intervals for Regression-Discontinuity Designs. *Econometrica*, 82(6):2295–2326.
- Cantoni, D., Yang, D. Y., Yuchtman, N., and Zhang, Y. J. (2019). Protests as Strategic Games: Experimental Evidence from Hong Kong's Antiauthoritarian Movement. Quarterly Journal of Economics, 134(2):1021–1077.
- Cattaneo, M. D., Jansson, M., and Ma, X. (2020). Simple Local Polynomial Density Estimators. Journal of the American Statistical Association, 115(531):1449–1455.
- Conley, T. G. (1999). GMM Estimation with Cross-sectional Dependence. *Journal of Econometrics*, 92(1):1–45.
- Crabtree, C., Kern, H. L., and Pfaff, S. (2018a). Data for: Mass Media and the Diffusion of Collective Action in Authoritarian Regimes: The June 1953 East German Uprising. *International Studies Quarterly*, 62(2):301–314. https://doi.org/10.1093/isq/sqy007 (accessed October 12, 2021).
- Crabtree, C., Kern, H. L., and Pfaff, S. (2018b). Mass Media and the Diffusion of Collective Action in Authoritarian Regimes: The June 1953 East German Uprising. *International Studies Quarterly*, 62(2):301–314.

- Dehdari, S. and Gehring, K. (2019). The Origins of Common Identity: Evidence From Alsace-Lorraine. Forthcoming at American Economic Journal: Applied Economics.
- Dell, M. (2010). The Persistent Effects of Peru's Mining Mita. Econometrica, 78(6):1863–1903.
- Dell, M. and Querubin, P. (2018). Nation Building Through Foreign Intervention: Evidence from Discontinuities in Military Strategies. *Quarterly Journal of Economics*, 133(2):701–764.
- Department of Defense (2016). Theatre History of Operations Reports (THOR): World WAR II dataset. United States Department of Defense. https://data.world/datamil/world-war-ii-thor-data (accessed October 11, 2021).
- Eder, C. and Halla, M. (2018). On the Origin and Composition of the German East-West Population Gap. IZA Working Paper 12031.
- Enikolopov, R., Makarin, A., and Petrova, M. (2020). Social Media and Protest Participation: Evidence From Russia. *Econometrica*, 88(4):1479–1514.
- Falter, J. W. (1997). The State Parliament Elections of 1946 in the Soviet-Occupied Zone. GESIS Data Archive, Cologne. ZA2879 Data file Version 1.0.0, https://doi.org/10.4232/1.2879.
- Falter, J. W. and Hänisch, D. (1990). Wahl- und Sozialdaten der Kreise und Gemeinden des Deutschen Reiches von 1920 bis 1933. GESIS Data Archive, Cologne. ZA8013 Data file Version 1.0.0, https://doi.org/10.4232/1.8013.
- Ferwerda, J. and Miller, N. (2014). Political Devolution and Resistance to Foreign Rule: A Natural Experiment. American Political Science Review, 108(3):642–660.
- Fisher, R. A. (1935). The Design of Experiments. Oliver and Boyd.
- Fontana, N., Nannicini, T., and Tabellini, G. (2017). Historical Roots of Political Extremism: The Effects of Nazi Occupation of Italy. CEPR Working Paper 11758.
- Freedom House (2020). Freedom in the World 2020. A Leaderless Struggle for Democracy.
- Fuchs-Schündeln, N. and Masella, P. (2016). Long-Lasting Effects of Socialist Education. The Review of Economics and Statistics, 98(3):428–441.
- Gagliarducci, S., Onorato, M., Sobbrio, F., and Tabellini, G. (2020). War of the Waves: Radio and Resistance during World War II. American Economic Journal: Applied Economics, 12(4):1–38.
- Gebhardt, M. (2017). Crimes Unspoken: The Rape of German Women at the End of the Second World War. Polity, Cambridge, UK.
- Gilbert, M. (2008). The Routledge Atlas of the Second World War. Routledge, New York, NY.
- Goebel, J., Grabka, M. M., Liebig, S., Kroh, M., Richter, D., Schröder, C., and Schupp, J. (2019). The German Socio-Economic Panel (SOEP). Jahrbücher für Nationalökonomie und Statistik, 239(2):345–360.
- Gregory, P. R. (2009). Terror by Quota: State Security from Lenin to Stalin (an Archival Study). Yale University Press.

Grieder, P. (2012). The German Democratic Republic. Palgrave Macmillan, West Sussex, UK.

- Grossmann, J., Juradja, S., and Roesel, F. (2021). Forced Migration, Staying Minorities, and New Societies: Evidence from Post-war Czechoslovakia. IZA Working Paper 14191.
- Hänisch, D. (1989). Inhalt und Struktur der Datenbank Wahl- und Sozialdaten der Kreise und Gemeinden des Deutschen Reiches von 1920 bis 1933. Historical Social Research.
- Henke, K.-D. (2009). Die amerikanische Besetzung Deutschlands. Walter de Gruyter.
- Henry, N. (2011). War and Rape: Law, Memory and Justice. Routledge, New York, NY.
- Imbens, G. W. and Lemieux, T. (2008). Regression Discontinuity Designs: A Guide to Practice. Journal of Econometrics, 142(2):615–635.
- Iyer, L. and Mani, A. (2012). Traveling Agents: Political Change and Bureaucratic Turnover in India. Review of Economics and Statistics, 94(3):723–739.
- Jones, M. (2015). After Hitler: The Last Ten Days of World War II in Europe. New American Library, New York, NY.
- Kahneman, D. and Miller, D. (1986). Norm Theory: Comparing Reality to its Alternatives. Psychological Review, 93:136–153.
- Kahneman, D. and Tversky, A. (1979). Prospect Theory: An Analysis of Decision under Risk. *Econometrica*, 47(2):263–291.
- Kern, H. L. and Hainmueller, J. (2009). Opium for the Masses: How Foreign Media Can Stabilize Authoritarian Regimes. *Political Analysis*, 17(4):377–399.
- Kocher, M. A., Pepinsky, T. B., and Kalyvas, S. N. (2011). Aerial Bombing and Counterinsurgency in the Vietnam War. American Journal of Political Science, 55(2):201–218.
- Kopstein, J. (1996). Chipping Away at the State: Workers' Resistance and the Demise of East Germany. World Politics, 48(3):391–423.
- Kowalczuk, I.-S. (2003). 17 Juni 1953–Volksaufstand in der DDR Ursachen, Abläufe, Folgen. Edition Temmen.
- Laudenbach, C., Malmendier, U., and Niessen-Ruenzi, A. (2020). The Long-lasting Effects of Living under Communism on Attitudes towards Financial Markets. NBER Working Paper 26818.
- Lichter, A., Löffler, M., and Siegloch, S. (2021a). Data for: The Long-Term Costs of Government Surveillance: Insights from Stasi Spying in East Germany. *Journal of the European Economic As*sociation, 19(2):741–789. https://doi.org/10.7910/DVN/WMVC4U (accessed October 12, 2021).
- Lichter, A., Löffler, M., and Siegloch, S. (2021b). The Long-Term Costs of Government Surveillance: Insights from Stasi Spying in East Germany. Journal of the European Economic Association, 19(2):741–789.
- MacDonald, C. (1993). The Last Offensive: United States Army in World War II, The European Theater of Operations. Center of Military History, United States Army, Washington, D.C.
- Martinez, L. R., Jessen, J., and Xu, G. (2021). Replication Files for: A Glimpse of Freedom: Allied Occupation and Political Resistance in East Germany. American Economic Association [publisher], Inter-university Consortium for Political and Social Research [distributor].
- Martinez-Bravo, M., Mukherjee, P., and Stegmann, A. (2017). The Non-Democratic Roots of Elite Capture: Evidence From Soeharto Mayors in Indonesia. *Econometrica*, 85(6):1991–2010.
- Mohr, C. (2021). Carrots and Sticks: Targeting the Opposition in an Autocratic Regime. https://tinyurl.com/bj382n9a (accessed October 12, 2021).
- Müller-Enbergs, H., Wielgohs, J., Hoffmann, D., Herbst, A., and Kirschey-Feix, I. (2010). Wer war wer in der DDR?: ein Lexikon ostdeutscher Biographien. Links.
- Murphy, D., Kondrashev, S., and Bailey, G. (1997). *Battleground Berlin: CIA vs KGB in the Cold War*. Yale University Press.
- Naimark, N. (1997). The Russians in Germany: A History of the Soviet Zone of Occupation, 1945-1949. Belknap Press.
- NARA (1944). Germany 1944-1945. National Archives and Records Administration, Franklin D. Roosevelt Library (LP-FDR), ARC Identifier: 16618469.
- Natkiel, R. (1985). Atlas of World War II. Bison Books, Greenwich, CT.
- Ochsner, C. (2017). Dismantled Once, Diverged Forever? A Quasi-natural Experiment of the Red Army's Misdeeds in Post-WWII Europe. ifo Working Paper 240.
- Ochsner, C. and Roesel, F. (2020). Migrating Extremists. *Economic Journal*, 130(628):1135–1172.
- Olson, M. (1965). The Logic of Collective Action: Public Goods and the Theory of Groups. Harvard University Press.
- Pimlott, J. (1995). The Viking Atlas of World War II. Viking, London, UK.
- Pons, V. (2018). Will a Five-Minute Discussion Change Your Mind? A Countrywide Experiment on Voter Choice in France. American Economic Review, 108(6):1322–63.
- Rozenas, A. and Zhukov, Y. M. (2019). Mass Repression and Political Loyalty: Evidence from Stalin's 'Terror by Hunger'. American Political Science Review, 113(2):569–583.
- Rühle, J. and Holzweißig, G. (1988). 13. August 1961. Die Mauer von Berlin. Wissenschaft und Politik.
- Schneider, D. M. (1989). Renaissance und Zerstörung der kommunalen Selbstverwaltung in der sowjetischen Besatzungszone. Oldenbourg.
- Schröder, R. (2010). Volkskammer der DDR. Teil 2: 10. Wahlperiode. Bundesarchiv, Koblenz.
- Schulz, K. J. and Briskey, J. A. (2005). Major Mineral Deposits of the World: Open-File Report 2005-1294, U.S. Geological Survey. https://mrdata.usgs.gov/major-deposits/ geo-inventory.php (accessed October 11, 2021).

- Schumann, A. (2014). Persistence of Population Shocks: Evidence from the Occupation of West Germany after World War II. American Economic Journal: Applied Economics, 6(3):189–205.
- SOEP (2019). German Socio-Economic Panel (SOEP), Data for years 1984-2018, version 35. https://doi.org/10.5684/soep-core.v35 (accessed October 13, 2021).
- Stargardt, N. (2015). The German War: A Nation Under Arms, 1939-1945. Basic Books.
- Statistisches Bundesamt (1994). Ausgewählte Zahlen der Volks- und Berufszählungen und Gebäudeund Wohnungszählungen 1950-1981 - Heft 15. Statistisches Bundesamt, Wiesbaden.
- Statistisches Zentralamt (1948). Volks- und Berufszählung vom 29. Oktober 1946 in der sowjetischen Besatzungszone Deutschlands. Deutscher Zentralverlag, Berlin.
- Stegmann, A. (2019). When East Meets West: Interpersonal Contact and the Demand for Democracy. https://tinyurl.com/4ec7t4r8 (accessed October 13, 2021).
- Steiner, A. (2013). The Plans that Failed: An Economic History of the GDR. Berghahn Books.
- Stibbe, M. (2006). The SED, German Communism and the June 1953 Uprising: New Trends and New Research. In McDermott, K. and Stibbe, M., editors, *Revolution and Resistance in Eastern Europe: Challenges to Communist Rule.* Berg Publishers.
- Stivers, W. and Carter, D. (2017). The City Becomes a Symbol: The US Army in the Occupation of Berlin, 1945-1949. Center of Military History, US Army.
- Svolik, M. W. (2013). Learning to Love Democracy: Electoral Accountability and the Success of Democracy. American Journal of Political Science, 57(3):685–702.
- Tarrow, S. (1998). Power in Movement: Social Movements and Contentious Politics. Cambridge University Press.
- Toland, J. (2003). The Last 100 Days: The Tumultuous and Controversial Story of the Final Days of World War II in Europe. Modern Library, New York, NY.
- Tullock, G. (1971). The Paradox of Revolution. *Public Choice*, 11:89–99.
- U.S. Army (1945). Atlas of the World Battle Fronts in Semimonthly Phases: A Supplement to the Biennial Report of the Chief of Staff of the United States Army to the Secretary of War (July 1, 1943 to June 30, 1945). U.S. Army.
- Welsh, H. (2010). Revolutionärer Wandel auf Befehl?: Entnazifizierungs-und Personalpolitik in Thüringen und Sachsen (1945-1948), volume 58. Walter de Gruyter.
- Wood, E. J. (2006). Variation in Sexual Violence during War. Politics and Society, 34(3):307–341.
- Ziemke, E. (1976). The U.S. Army in the Occupation of Germany: 1944-1946. Center of Military History, United States Army, Washington, D.C.

Figures



Figure 1: Occupied counties in East Germany at the end of WWII (1945)

Notes: Blue indicates occupation by Allied Expeditionary Force, red shows occupation by the Soviet Union. Dates based on occupation of the largest city in a county (Kreis). County boundaries from January 1, 1948. The uncolored counties in the south on May 7 correspond to an area later known as the *Free Republic of Schwarzenberg*, which was not initially occupied by Allied or Soviet forces and only captured by Soviets on June 24, 1945. The city of Berlin is tagged with a star.



Figure 2: The Line of Contact and Protest Incidence in 1953

(a) Municipality-level assignment

(b) 1953 uprising

Notes: Map in Panel a shows state and municipal boundaries of the contemporary Federal Republic of Germany (as of December 31, 2016) in the area corresponding to the former German Democratic Republic (GDR). Municipalities in blue were initially occupied by the Allied Expeditionary Force, while those in red were initially captured by the Soviet Union. The uncolored counties in the south correspond to an area later known as the Free Republic of Schwarzenberg, which was not initially occupied by Allied or Soviet forces and only captured by Soviets on June 24, 1945. Panel b shows the location of municipalities with information on protest incidence during the 1953 uprising.





Notes: Units of observation are East German municipalities with a population above 1,000. Figure shows means of an indicator for the occurrence of protests during the 1953 uprising computed in 2km bins around the line of contact. Negative distances from the line denote the Allied side of the line of contact, positive distances correspond to the Soviet side. The fitted polynomials are based on regressions using unbinned data with either a linear (solid line), quadratic (dashed line) or cubic (long dashed line) polynomial of the forcing variable.

Figure 4: 1953 protests - RDD placebo estimates



Notes: Figure shows the distribution of 10,000 RDD coefficients obtained from randomly drawn lines dividing East Germany. The vertical black line indicates the estimated coefficient in column 5 of Table 3. See Table 3 for additional notes. *p*-value denotes the share of placebo estimates that are larger than the observed value. For illustration purposes, coefficients below -0.3 and above 0.3 are not displayed in the histogram (0.82% of estimates). Randomly-drawn lines are restricted to ensure that the Allied- and Soviet-assigned sides contain no more than 60% of observations each.





(b) Excess Turnover in Allied-occupied Cities

Notes: Panel a shows the month of appointment for mayors in Allied- and Soviet-occupied cities in 1945 (N = 327, 156 cities). Panel b shows point estimates and 95% confidence intervals from a regression of an excess turnover indicator (yearly appointments > 1) on the interaction of year dummies with a dummy for cities initially occupied by the Allies (N = 3, 356, 172 cities). Controls include city and year fixed effects. Standard errors clustered by city.



Figure 6: Perceptions of initial occupation and change in occupier's identity

(a) Initial occupation (Allied effect)



(b) Change of occupying army (Allied to Soviet)

Notes: All dependent variables are coded as binary indicators for ease of interpretation. The sample in Panel a is restricted to individuals whose stated location in May 1945 is within 100 km of the line of contact and that correctly identify the identity of the initial occupying army for the stated location. Coefficients in Panel a are obtained from a regression on an indicator of Allied occupation. Appendix C shows the detailed wording of the questions and Appendix Table A3 displays robustness of coefficients. In Panel b the sample consists of individuals in Allied-occupied areas (i.e. who experienced a change from Allied to Soviet occupation). Whiskers indicate 95% confidence intervals. Source: Own survey.

Figure 7: Difference in Communist vote share between West and East Berlin



Notes: Plot shows differences in the Communist vote share between West and East Berlin, relative to 1929 (omitted category). Vote share corresponds to the Communist Party of Germany (KPD) until 1932 and to the Socialist Unity Party (SED) in 1946. Units of observation are 75 districts of Berlin (40 in West Berlin). 1929 was a municipal election, 1930 and 1932 federal elections and 1946 the Berlin state election. Estimation includes district and year fixed effects (see Table 7, column 5 for the regression results). Standard errors are clustered at the district level. Whiskers indicate 95% confidence intervals.

Tables

Table 1:	: Pre-	determined	characteristics	in	Allied-	and	Soviet	-occupied	municipa	lities
----------	--------	------------	-----------------	----	---------	-----	--------	-----------	----------	--------

	Mean	Diff: Allied-Soviet		Obs.
	(1)	(2)	(3)	(4)
Panel A: Municipality level (pre-war, ≥ 2	,000 population))	()	
Log total population 1933	8 843	-0.139	0 243	519
Log total population 1555	0.040	(0.100)	(0.240)	015
Log working population 1933	9.156	-0.194	0.019	241
	0.200	(0.131)	(0.286)	
Female share of working population 1933	0.333	-0.007	-0.017	241
		(0.008)	(0.014)	
Share unemployed 1933	0.204	0.008	-0.007	241
		(0.010)	(0.019)	
Share of agricultural workers 1933	0.167	-0.009	0.006	241
		(0.020)	(0.037)	
Share of industrial workers 1933	0.364	0.038^{**}	0.012	241
		(0.019)	(0.023)	
Share protestant 1925	0.896	-0.004	-0.014	310
		(0.020)	(0.009)	
Share catholic 1925	0.062	0.007	0.008	310
		(0.020)	(0.009)	
Share Jewish 1925	0.004	-0.003*	0.001	226
		(0.002)	(0.001)	
Voter turnout 1933	0.912	0.012^{***}	-0.008	518
		(0.004)	(0.006)	
NSDAP share 1933	0.458	-0.009	-0.009	518
		(0.013)	(0.021)	
SPD share 1933	0.229	-0.017	0.001	518
		(0.012)	(0.025)	
KPD share 1933	0.162	0.038^{***}	0.005	518
		(0.011)	(0.019)	
Panel B: Municipality level (1946)				
Distance to Berlin (km)	160.213	76.256***	-1.761	5649
		(9.055)	(3.839)	
Distance to closest concentration camp (km)	75.189	-19.509***	-5.665	5649
- 、 /		(4.864)	(3.738)	
Distance to inner German border (km)	91.860	-90.698***	3.246	5649
		(7.301)	(4.583)	
Bombing within $10 \text{km} (0/1)$	0.445	0.245^{***}	-0.064	5649
		(0.040)	(0.108)	
Synagogue destroyed $(0/1)$	0.024	0.016^{*}	0.020	5649
		(0.009)	(0.023)	
Elevation (m)	163.117	112.143^{***}	-29.311*	5649
		(23.230)	(15.861)	
Distance to closest mineral reserve (km)	113.331	-88.472***	1.276	5649
		(13.352)	(4.938)	
Distance to closest coast (km)	229.683	91.821***	4.057	5649
		(18.313)	(5.039)	
Forcing variable $f(\mathbf{g}_i)$			$Linear \times Allied$	
Border segment FEs	_	_	Υ	
Bandwidth	Full	Full	$50 \mathrm{km}$	$50 \mathrm{km}$

Notes: Units of observation in Panel A are municipalities in East Germany with a population size above 2,000 during the Weimar Republic. Units of observation in Panel B are municipalities in East Germany in 1946 (corresponding to the 1946 election and census outcomes). Column 1 shows the sample mean, column 2 shows the mean difference between Allied- and Soviet-occupied areas. Column 3 shows the difference after the inclusion of the forcing variable and ten border segment FEs in a sample window of 50 km around the line of contact. Column 4 shows the number of observations for the 50 km bandwidth. Labor force is defined as all individuals that were either self-employed, civil servants, employees, workers or domestic workers. Online Appendix A provides a detailed description of the data sources and variables used. Standard errors clustered at the county level in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1

	Dependent variable: Days of Allied occupation (Mean: 42.6)							
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
Allied	$74.724^{***} \\ (1.400) \\ [1.394]$	71.369*** (2.118) [2.109]	$71.858^{***} \\ (1.905) \\ [1.896]$	$72.164^{***} \\ (2.064) \\ [2.052]$	$73.328^{***} \\ (1.711) \\ [1.700]$	$72.989^{***} \\ (2.512) \\ [2.496]$	$72.147^{***} \\ (0.267)$	
Forcing variable $f(\mathbf{g}_i)$		Linear	L	inear \times Allie	ed	Quadratic \times All.	Local poly.	
Border segment FEs Full controls				Y	Y Y	Y Y		
Bandwidth Observations	$50 \mathrm{km}$ 5,649	$50 \mathrm{km}$ 5,649	$50 \mathrm{km}$ 5,649	$50 \mathrm{km}$ 5,649	$50 \mathrm{km}$ 5,649	$50 \mathrm{km}$ 5,649	$50 { m km} \\ 5,649$	

Table 2: Exposure to Allied forces around the Line of Contact

Notes: Units of observation are East German municipalities in 1946. Dependent variable is the number of days of Allied occupation. Exposure is calculated using (i) the occupation date of the largest city in the county (*Kreis*) in which the municipality is located and (ii) July 3 as the date of Allied withdrawal. Estimate in column 7 based on Calonico et al. (2014). Border segment FEs correspond to ten equally large segments of the distribution of latitudes of municipalities in the 50 km bandwidth around the line of contact. Controls are latitude, longitude and all municipality-level variables of Table 1 (Panel B). Standard errors clustered at the county level in parentheses. Conley standard errors in square brackets. *** p < 0.01, ** p < 0.05, * p < 0.1

		Deper	ndent vari	able: Incide	ence of prot	tests (Mean: 0.22)		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
Allied	0.081**	0.163***	0.142**	0.143***	0.170***	0.182**	0.150**	
	(0.033)	(0.057)	(0.058)	(0.051)	(0.061)	(0.085)	(0.072)	
	[0.026]	[0.059]	[0.058]	[0.056]	[0.061]	[0.088]		
Forcing variable $f(\mathbf{g}_i)$		Linear	L	$inear \times All$	lied	Quadratic \times All.	Local poly.	
Border segment FEs				Y	Y	Υ		
Full controls					Υ	Υ		
Sample				Population $\geq 1,000$				
Bandwidth	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	
Observations	968	968	968	968	968	968	968	

Table 3:	Incidence of	protests	during	the	1953	Uprising	around	the	Line	of	Contact
----------	--------------	----------	--------	-----	------	----------	--------	-----	------	----	---------

Notes: The dependent variable is an indicator for the occurrence of protests during the 1953 uprising. The sample consists of all East German municipalities with a population above 1,000. Estimate in column 7 based on Calonico et al. (2014). Border segment FEs correspond to ten equally large segments of the distribution of latitudes of municipalities in the 50 km bandwidth around the line of contact. Controls are latitude, longitude and all municipality-level variables of Table 1 (Panel B). Standard errors clustered at the county level in parentheses. Conley standard errors in square brackets. *** p < 0.01, ** p < 0.05, * p < 0.1

Panel A: Main estimation sample		Der er de		le. Tereide			
		Depende	ent variab.	le: Inclue	nce or prot	ests	
Mean of dep. variable	0.22	0.22	0.13	0.13	0.27	0.27	
	(1)	(2)	(3)	(4)	(5)	(6)	
Allied	0 1/9**	0 170***	0.150*	0.148	0.120*	0 137*	
Amed	(0.142)	(0.061)	(0.130)	(0.140)	(0.120)	(0.080)	
	(0.058)	[0.001]	(0.077)	(0.113) [0, 101]	(0.008) [0.070]	(0.080)	
	[0.058]	[0.001]	[0.077]	[0.101]	[0.079]	[0.031]	
Line:	Ful	l line	La	nd]	Rivers	
Forcing variable $f(\mathbf{g}_i)$			Linea	$ar \times Allied$			
Border segment FEs		Y		Y		Y	
Full Controls		Υ		Υ		Υ	
Sample			Popula	tion > 1 .	000		
Bandwidth	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	
Observations	968	968	366	366	602	602	
Danal D. Dlaasha waang		Den en de		le. Tereide			
Panel B: Placebo rivers		Depende	ent variab	le: Incide	nce of prot	ests	
Mean of dep. variable	0.25	0.25	0.18	0.18	0.26	0.26	
	(1)	(2)	(3)	(4)	(5)	(6)	
Allied	-0.021	-0.002	-0.002	-0.021	-0.006	-0.021	
, inica	(0.021)	(0.059)	(0.002)	(0.021)	(0.065)	(0.021)	
	[0.061]	[0.064]	[0.065]	[0.066]	[0.067]	[0.066]	
	[0.000]	[0.001]	[0.000]	[0.000]	[0.001]	[0.000]	
Line:	Saale (p	olacebo 1)	Elbe (pl	acebo 2)	Spree-Ha	vel (placebo 3)	
Forcing variable $f(\mathbf{g}_i)$			Linea	$ar \times Allie$	d		
Border segment FEs		Y		Y		Y	
Full Controls		Υ		Υ		Υ	
Sample			Popula	tion > 1 .	000		
Bandwidth	$50 \mathrm{km}$	$50 \mathrm{km}$	50km	50km	$50 \mathrm{km}$	$50 \mathrm{km}$	
Observations	787	787	530	530	579	579	

Table 4: Incidence of protest along the *Line of Contact* - main estimates, land, river and placebo lines

Notes: The dependent variable is an indicator for the occurrence of protests during the 1953 uprising. The sample consists of all East German municipalities with a population above 1,000. Columns 1-2 of Panel A replicate estimates from Table 3. Columns 3-6 differentiates by whether the line of contact was defined by land or rivers. Panel B shows estimates from three placebo lines of contact running along rivers (see Appendix Figure A5). Border segment FEs correspond to ten equally large segments of the distribution of latitudes of municipalities in the 50 km bandwidth around the line of contact. Controls are latitude, longitude and all municipality-level variables of Table 1 (Panel B). Standard errors clustered at the county level in parentheses. Conley standard errors in square brackets. *** p < 0.01, ** p < 0.05, * p < 0.1

	ln Total		Share					
	Pop.	Female	Protestant	Agr.	Ind.	Svc. & Trade	Protest	s 1953
Mean of dep. variable	6.27	0.56	0.83	0.47	0.23	0.10	0.22	0.22
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Allied	$\begin{array}{c} 0.222 \\ (0.146) \\ [0.073] \end{array}$	-0.000 (0.003) [0.003]	-0.012 (0.009) [0.008]	-0.072** (0.035) [0.030]	0.033 (0.023) [0.019]	$\begin{array}{c} 0.048^{**} \\ (0.020) \\ [0.017] \end{array}$	$\begin{array}{c} 0.170^{***} \\ (0.061) \\ [0.061] \end{array}$	$\begin{array}{c} 0.148^{**} \\ (0.060) \\ [0.059] \end{array}$
Bandwidth	50km	50km	50km	50km	50km	50km	50km	50km
Forcing variable $f(\mathbf{g}_i)$				Linear	\times Allied			
Border segment FEs	Y	Y	Y	Y	Y	Y	Y	Y
Full controls	Υ	Υ	Υ	Υ	Υ	Υ	Υ	Υ
Sample		А	ll municipali	Pop. \geq	1,000			
Census 1946 controls							-	Υ
Observations	$5,\!648$	$5,\!648$	$5,\!648$	$5,\!648$	$5,\!648$	$5,\!648$	968	968

 Table 5: Testing for demographic changes around the line of contact

Notes: Units of observation in columns 1-6 are all East German municipalities of 1946. Dependent variables are log population, share female, share protestant and shares in agriculture, industry, service and trade, according to the 1946 census in the Soviet occupation zone. Units of observation in columns 7-8 are East German municipalities with a population over 1000. Border segment FEs correspond to ten equally large segments of the distribution of latitudes of municipalities in the 50 km bandwidth around the line of contact. Controls are latitude, longitude and all municipality-level variables of Table 1 (Panel B). Census 1946 controls are all dependent variables of columns 1-6. Regressions in columns 2-6 are population weighted. Standard errors clustered at the county level in parentheses. Conley standard errors in square brackets. *** p < 0.01, ** p < 0.05, * p < 0.1

	Protest	Log hou	sing built	Spying	Political
	1953	Pre 1946	Post 1946	intensity	arrests
Mean of dep. var	0.344	9.867	8.299	0.098	0.586
	(1)	(2)	(3)	(4)	(5)
Allied	0.362^{***}	0.373	0.447	0.035	0.213
	(0.126)	(0.246)	(0.311)	(0.022)	(0.153)
Forcing variable $f(\mathbf{g}_i)$		Li	$near \times Allie$	ed	
Border segment FEs	Y	Y	Y	Y	Υ
Full controls	Υ	Υ	Υ	Υ	Υ
Bandwidth	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$
Observations	85	85	85	77	83

Table 6: Public good provision and governance during the GDR

Notes: Units of observation are counties of the GDR. Protest denotes the share of municipalities in a county with protests in 1953. Data on housing built comes from the GDR housing census of 1971. Spying intensity (the number of unofficial informers divided by population) is for 1980-88; political arrests per capita is for 1984-88. Data on spying and political arrests from Lichter et al. (2021a). Border segment FEs correspond to ten equally large segments of the distribution of latitudes of counties. Controls are latitude, longitude and all variables of Table 1 (Panel B). Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1

Dependent variable:	S	ED vote sha	re	SED	/KPD vote	share
Mean of dep. var.	0.206	0.206	0.182	0.209	0.237	0.224
	(1)	(2)	(3)	(4)	(5)	(6)
Allied	-0.176***	-0.136***	-0.134***			
	(0.012)	(0.018)	(0.021)			
Allied \times 1930	(0.01-)	(0.010)	(0.0)		-0.002	0.002
					(0.009)	(0.010)
Allied \times 1932					-0.013	-0.026
					(0.013)	(0.020)
Allied \times 1946				-0.104^{***}	-0.105^{***}	-0.099***
				(0.019)	(0.018)	(0.021)
				V	V	V
Election year FE				Ŷ	Ŷ	Ŷ
District FE				Y	Y	Y
Latitude / longitude		Y	Y			
Common longitude support			Υ			Υ
Election sample:						
1946 Berlin state election	Υ	Υ	Υ	Υ	Υ	Υ
1929 municipal elections				Υ	Υ	Υ
1930 and 1932 federal elections					Υ	Y
Observations	75	75	49	126	247	164

Table 7: Difference in Communist vote share between West and East Berlin

Notes: Units of observation are districts in Berlin. Columns 1-3 show the difference in vote share of the Socialist Unity Party (SED) in the 1946 State election in between East (Soviet sector) and West Berlin (American, British and French sectors). Columns 4-6 include pre-war elections in a difference-indifferences setting using the vote share of the Communist Party of Germany (KPD), the predecessor of the SED. Column 4 restricts pre-war elections to the 1929 local election, while columns 5-6 additionally include federal elections in 1930 and 1932. Common longitudinal support indicates that sample only includes districts in West Berlin as far West as the westernmost district in East Berlin and vice versa. Standard errors (robust in columns 1-3, clustered at the district level in columns 4-6) in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Online Appendix

A GLIMPSE OF FREEDOM: ALLIED OCCUPATION AND POLITICAL RESISTANCE IN EAST GERMANY

Authors: Luis R. Martinez (r), Jonas Jessen and Guo Xu

Table of Contents

Appendix A Data Appendix	Appendix p.2
Appendix B Additional Figures and Tables	Appendix p.6
Appendix C Retrospective Survey	Appendix p.25

A Data Appendix

Establishing the line of contact

To identify the line that separates the area occupied by the Western Allies within East Germany at the end of WWII from that occupied by the Red Army, the line of contact, we consulted several sources. We studied the following accounts of the final days of the war: Ambrose (2000); Beevor (2002); Jones (2015); MacDonald (1993); Toland (2003). Particularly helpful was MacDonald (1993), as it not only contains a detailed account of the military moves of the allies, but it also includes a series of maps that document specific operations and an appendix with what appear to be official maps tracking the frontline at various locations and points in time. We extracted information from the daily communiqués by the Supreme Headquarters Allied Expeditionary Force (SHAEF), which provide daily updates on allied operations in the western front.⁴⁵ In some cases, we additionally conducted web searches of specific military units that seemed to have been present in areas where the other sources proved unsatisfactory.⁴⁶ Finally, we also used the following Atlases of the second world war: U.S. Army (1945); Natkiel (1985); Pimlott (1995); Badsey (2000); Gilbert (2008).

For the most part, all sources agree on the location of the line of contact. From north to south, it starts at Wismar on the Baltic and goes down Mecklenburg-Vorpommen (through Schwerin and Ludwiglust) up to Domitz on the Elbe. It then follows the Elbe from Domitz up to Dessau-Roßlau at the intersection with the Mulde and follows the Mulde up to Lunzenau. The last segment (stretching from Lunzenau to a point on the Czech border) was the most challenging to pin down, as it is either not shown in some maps (e.g. Pimlott, 1995; Badsey, 2000) or includes too few reference points, such as cities (e.g. U.S. Army, 1945; Natkiel, 1985). Our reconstruction of this segment is mainly based on the location of the easternmost cities in which the SHAEF communiqués confirmed allied presence. Additionally, we conducted desk research by consulting accounts of contemporary witnesses and other sources such as city websites indicating the identity of the occupying force to pin down the easternmost Allied occupied cities in this segment. From Lunzenau to the Czech border, those are Limbach-Oberfrohna, Lichtenstein, Wilkau-Haßlau, Lichteinstein, Auerbach (Vogtland) and Klingenthal. Having established the location of the line of contact, we assigned the identity of the initial occupying force at the municipality-level. Our main outcomes are all based on this fine-grained assignment of the "treatment". Migration patterns in the 1946 census

⁴⁵Consulted in March, 2020 at https://lib.byu.edu/collections/eisenhower-communiques

⁴⁶For example, the document "The Mass Surrender of German Troops to the 347th Infantry Regiment on May 6, 1945," consulted in March 2020 at http://87thinfantrydivision.com/tom-stafford/ the-mass-surrender-of-german-troops-to-the-347th-infantry-regiment-on-may-6-1945 was very useful in establishing allied presence in the state of Saxony.

(Appendix Table A4 and Appendix Table A5) are only availability at the county-level. As the line of contact played no role in administrative boundaries defined afterwards, a few counties are crossed by the line (predominantly in Saxony, where larger parts of the line are not defined by rivers). We assign these counties to be either Allied or Soviet occupied according to the predominant area or population share assigned to either side. We check the robustness of the results to the exclusion of these "divided" counties.

Data Sources

Predetermined characteristics - county level and larger municipalities: Unit of observation: Counties and municipalities with population above 2,000 (1925, 1933). Number of observations: 226-519 within 50km of line of contact. Differences stem from the different levels of aggregation across variables (see Hänisch, 1989). For example, some indicators were only aggregated by the statistical office for municipalities with a population above 10,000. Description: Official statistics from the German Reich based on population censuses in 1925 and 1933, and elections in 1933. Source: Falter and Hänisch (1990). Access conditions: Free, but GESIS login required. Outputs: Table 1 (panel A).

Predetermined characteristics - municipality level: Unit of observation: Municipalities (1946). Number of observations: 5,649 within 50km of line of contact. Description: WWII measures and time-invariant geographic characteristics. Sources: Jewish Virtual Library on Nazi concentration camps ⁴⁷, Synagogue Memorial "Beit Ashkenaz" on synagogues destroyed in 1938 Pogroms ⁴⁸, Department of Defense (2016) on WWII bombing (THOR dataset), Schulz and Briskey (2005) on closest mineral reserve. Own calculations for other variables. Access conditions: Free, but data.world login required for THOR dataset. Outputs: Table 1 (panel B), Online Appendix Figure A2.

Occupation date and length of exposure to Allied forces (1945): Unit of observation: Largest city per county in 1946. Number of observations: 178. Description: Date of occupation at the end of WWII. Change of occupation force on July 3, 1945. Source: Own data collection. Access conditions: Freely available from Martinez et al. (2021). Outputs: Figure 1, Online Appendix Figure A3, Table 2.

Protest indicator (1953): Unit of observation: Municipalities with population above 1000 in 1953. Number of observations: 968 within 50km of line of contact. Description: Dummy for occurrence of protests during 1953 uprising. Source: Crabtree et al. (2018b), based on Kowalczuk (2003). Access conditions: Freely available from Crabtree et al. (2018a).

⁴⁷Available at https://tinyurl.com/w442yx6 (accessed October 11, 2021).

⁴⁸Available at https://tinyurl.com/rmbxjb7k (accessed October 11, 2021).

Outputs: Figures 2-4, Online Appendix Figures A6-A9, Tables 3-5, Online Appendix Table A1, Online Appendix Table A7.

Soviet Occupation zone - Population census at municipality level (1946): Unit of observation: Municipalities in 1946. Number of observations: 5,649 within 50km of line of contact. Description: Demographic characteristics based on the 1946 population census in the Soviet occupation zone. Source: Falter (1997). Access conditions: GESIS login required, data released for academic research and teaching. Outputs: Table 5.

Soviet Occupation zone - Population census at county level (1946): Unit of observation: Counties in 1946. Number of observations: 178 (89 within 50km of line of contact). Description: Migration outcomes based on the 1946 population census in the Soviet occupation zone. Source: Own digitization from Statistisches Zentralamt (1948). Access conditions: Freely available from Martinez et al. (2021). Outputs: Online Appendix Tables A4-A5.

Soviet Occupation zone - State elections (1946): Unit of observation: Municipalities in 1946. Number of observations: 5,649 within 50km of line of contact. Description: Election outcomes (turnout, vote shares) from the 1946 state elections in the Soviet occupation zone. Source: Falter (1997). Access conditions: GESIS login required, data released for academic research and teaching. Outputs: Online Appendix Table A8.

Berlin districts - Election results (pre-WWII and 1946): Unit of observation: Berlin districts (1929, 1930, 1932, 1946). Number of observations: 247 in the panel, 75 in 1946. Description: KPD vote share in pre-war municipal and federal elections, SED vote share in 1946 Berlin state elections. Source: Department for Statistics Berlin-Brandenburg.⁴⁹ Access conditions: Free. Outputs: Figure 7, Table 7.

Municipal mayors (1945-1953): Unit of observation: Cities with population above 10,000 in 2020. Number of observations: 3,356 mayors from 172 cities. Description: Name, party affiliation and term dates of municipal mayors. Source: Own data collection. Access conditions: Freely available from Martinez et al. (2021). Outputs: Figure 5, Online Appendix Figure A10.

Housing stock (1971): Unit of observation: Counties in 1971 (GDR). Number of observations: 85 within 50km of line of contact. Description: Housing stock, disaggregated by date of construction. Source: Own digitization from Statistisches Bundesamt (1994). Access conditions: Freely available from Martinez et al. (2021). Outputs: Table 6.

Spying intensity and political arrests (1980s): Unit of observation: Counties in 1980 (GDR). Number of observations: 83 within 50 km of line of contact (77 with data on spying). Description: Number of Stasi spies and political arrests (normalized by population).

⁴⁹Available at https://tinyurl.com/4z4e89dw (accessed October 11, 2021.).

Source: Lichter et al. (2021a). Access conditions: Free. Outputs: Table 6.

Protests (1989): Unit of observation: Municipalities with population above 1,000 in 1989. Number of observations: 900 within 50km of line of contact. Description: Daily information on occurrence of protest events and additional information on their characteristics. Source: Archiv Bürgerbewegung.⁵⁰ Access conditions: Free. Outputs: Online Appendix Table A6.

Election results (1990): Unit of observation: Municipalities in 1990. Number of observations: 2,262. Description: PDS and Allianz für Deutschland vote share in 1990 election, aggregated from polling-station level. Source: Own digitization of Schröder (2010), based on German Federal Archives. Access conditions: Freely available from Martinez et al. (2021). Outputs: Online Appendix Table A6.

Socio-economic characteristics (2018): Unit of observation: Individuals living in East Germany in 2018 and born before 1960. Number of observations: 2,183. Description: Survey responses in German Socio-economic Panel (SOEP, v35). Source: SOEP (2019). See Goebel et al. (2019) for data documentation. Access conditions: Application and data use agreement required.⁵¹ Outputs: Online Appendix Table A2.

Survey responses (2020): Unit of observation: Individuals living in East Germany in 2020 and born before 1960. Number of observations: 472 on initial occupation, 216 on change of occupying army. Description: Survey responses on own or parents' occupation experience at the end of WWII. Source: Own data collection. Access conditions: Freely available from Martinez et al. (2021). Outputs: Figure 6, Online Appendix Figure A11, Online Appendix Table A2, Online Appendix Table A3.

⁵⁰Available at https://tinyurl.com/5ex6t8za (accessed on October 13, 2021).

⁵¹Available at https://tinyurl.com/e9rz8pvz (accessed on October 13, 2021).

B Additional Figures and Tables

Figure A1: The Times, June 7 1945

RUSSIAN MAP OF ZONES OF OCCUPATION

ALL THURINGIA AND SAXONY = IN SOVIET SPHERE

LARGE AREA IN BRITISH AND U.S. HANDS TO BE GIVEN UP

BOUNDARIES IN AUSTRIA

A map of the allied zones of occupation in Germany published in Russian newspapers yesterday indicated that the whole of the German States of Thuringia, Saxony, Mecklenburg, and Anhalt, as well as the Prussian province of Saxony, would fall on the Russian side of the line of demarcation.

The Moscow announcement has been received with some surprise in London, for, though it is not questioned that the Russian zone is substantially as shown, it is felt that the statement should have been a joint one and that it would have come more fittingly from the Control Council in Berlin.

(a)

Notes: Excerpts from The Times newspaper cover from June 7 1945 showing the cover title and a detailed map of the agreed upon occupation zones.



(b)



Figure A2: Density of observations around the line of contact

Notes: Figure plots the density of 1946 municipalities around the line of contact using the estimator proposed in Cattaneo et al. (2020). Negative numbers denote the Allied side of the line and positive numbers the Soviet-occupied side, respectively. For estimation a cubic polynomial and a symmetric bandwidth of 50km is used.



Figure A3: Duration of Allied occupation around the *Line of Contact* (1945)

Notes: Units of observation are municipalities in East Germany in 1946. Plot shows raw means of the number of days of exposure to Allied forces computed in 2 km bins around the line of contact. The fitted polynomials are based on regressions using unbinned data with a linear (solid line), quadratic (dashed line) and cubic (long dashed line) polynomial. Negative numbers denote the Allied side of the line and positive numbers correspond to the Soviet side. Exposure is calculated using (i) the occupation date of the largest city in the county (*Kreis*) in which the municipality is located and (ii) July 3 as the date of Allied withdrawal.



Figure A4: Placebo divisions of East Germany

Notes: Figure shows four illustrative placebo divisions of East Germany on which the estimates in Figure 4 are based. Blue dots denote Allied assigned municipalities, red dots Soviet assigned ones. Restriction on random divisions are that each side may contain at most 60% of municipalities.



Notes: Map shows rivers in East Germany. Blue colored municipalities were initially occupied by the Allies, red colored ones were captured by the Soviet Union. Dark blue lines show the largest rivers flowing through East Germany. The highlighted river segments denote the river placebo lines used in Table 4.



Figure A6: 1953 protest result with border segments omitted

Notes: Figure shows regression estimates corresponding to Table 3 where each of the ten equally large border segments are dropped in turn. Estimates are based on the full specification in column 5. Square denotes the reference estimate of Table 3. Whiskers show 95% confidence intervals.



Figure A7: 1953 protest results with states omitted

Notes: Figure shows regression estimates corresponding to Table 3 where each of the five states are dropped in turn. BB Brandenburg, MV Mecklenburg-Vorpommern, SA Saxony, ST Saxony-Anhalt, TH Thuringia. Estimates are based on the full specification in column 5. Square denotes the reference estimate of Table 3. Whiskers show 95% confidence intervals.



Figure A8: 1953 protest with varying bandwidth

Notes: Figure shows regression estimates of Table 3 with varying bandwidths. Regressions are based on the full specification in column 5. The black square at 50 km indicates the coefficient displayed in the table, the dark grey diamond shows the estimate based on the data-driven bandwidth selection using the procedure developed by Calonico et al. (2014). Dark blue lines shows 95% confidence intervals.

Figure A9: 1953 protest results with municipalities dropped in vicinity to the line



Notes: Figure shows regression estimates corresponding to Table 3 where subsequently municipalities within x km to the line are dropped (*donut RDD*). Estimates are based on the full specification in column 5. Square denotes the reference estimate of Table 3. Whiskers show 95% confidence intervals.

Figure A10: End of tenure of Allied appointed mayors by party affiliation



Notes: Figure shows the year in which the tenure of mayors appointed under Allied occupation (March-June 1945) ended, disaggregated by party affiliation (N = 42). Left-wing parties are KPD and SPD. Other includes CDU, DDP, LDP, NSDAP, and those without partian affiliation. Mayors without recorded affiliation are dropped in this exposition.



Figure A11: Attrition of survey sample

Notes: Figure shows the attrition of the survey sample. The starting sample exceeds the number of respondents (N = 1, 002) as second-hand accounts can come from mothers and fathers if the occupation experience was different. The left hand-side of the figure in light-blue corresponds to Panel a of Figure 6, the red right-hand side to Panel b. Source: Own survey.

		Depen	dent varia	ble: Incide	nce of pro	tests (Mean: 0.21)			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)		
Allied	0.080^{**}	0.142^{***}	0.131^{**}	0.119^{***}	0.130^{**}	0.128^{*}	0.137^{**}		
	(0.031)	(0.053)	(0.053)	(0.045)	(0.055)	(0.073)	(0.064)		
	[0.025]	[0.052]	[0.051]	[0.050]	[0.054]	[0.076]	· · ·		
Forcing variable $f(\mathbf{g}_i)$		Linear	Li	$near \times Alli$	ed	Quadratic \times All.	Local poly.		
Border segment FEs			-	Y	Y	Y			
Full controls					Υ	Υ			
Sample				Population $\geq 1,000$					
Bandwidth	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$		
Observations	1,043	1,043	1,043	1,043	1,043	1,043	1,043		

Table A1: Incidence of protests during the 1953 Uprising around the *Line of Contact* including Free Republic of Schwarzenberg

Notes: The dependent variable is an indicator for the occurrence of protests during the 1953 uprising. The sample consists of all East German municipalities with a population above 1,000, including municipalities of the initially unoccupied *Free Republic of Schwarzenberg*. These observations are assigned to be Soviet-occupied, as Soviet troops captured the area on June 24, 1945. Estimate in column 7 based on Calonico et al. (2014). Border segment FEs correspond to ten equally large segments of the distribution of latitudes of municipalities in the 50 km bandwidth around the line of contact. Controls are latitude, longitude and all municipality-level variables of Table 1 (Panel B). Standard errors clustered at the county level in parentheses. Conley standard errors in square brackets. *** p < 0.01, ** p < 0.05, * p < 0.1

	Own survey	SOEP
	(1)	(2)
Age in years	67.89	72.73
	(6.39)	(8.77)
High education	0.48	0.18
	(0.50)	(0.38)
Female	0.41	0.55
	(0.49)	(0.50)
Birth year of mother	1,923.83	1,923.92
	(8.65)	(8.76)
Current state of residence	_	
Berlin	0.19	0.18
	(0.39)	(0.38)
Brandenburg	0.21	0.17
	(0.41)	(0.37)
Mecklenburg-Vorpommern	0.07	0.10
	(0.25)	(0.30)
Saxony	0.33	0.26
	(0.47)	(0.44)
Saxony-Anhalt	0.09	0.15
	(0.29)	(0.36)
Thuringia	0.11	0.13
	(0.31)	(0.34)
Observations	472	2183

 Table A2:
 Comparison of survey sample with East German population

Notes: Table compares the survey sample from Figure 6 with the East German population as of 2018. Both samples are restricted to individuals living in East Germany born before 1960. High education denotes that respondents have obtained at least upper secondary education. Birth year of mother only shown for individuals born after 1940 (see Appendix C). Means in SOEP calculated using sample weights, making the sample representative for the population. Source: Own survey and SOEP v35

	Sample mean	Allied coefficient			Ν	
	(1)	(2)	(3)	(4)	(5)	(6)
Arrival of occupying army perceived positively	0.233	0.295***	0.295***	0.261***	0.265***	391
		(0.054)	(0.053)	(0.062)	(0.061)	
Occupying army distributed food	0.666	0.143^{**}	0.124^{**}	0.144^{**}	0.121^{*}	317
		(0.056)	(0.056)	(0.063)	(0.064)	
Occupying army distributed medicine / cared for injured	0.595	0.158^{**}	0.145^{**}	0.169^{**}	0.149**	259
		(0.066)	(0.066)	(0.076)	(0.074)	
Occupying army mediated conflicts	0.637	0.063	0.068	0.047	0.054	234
		(0.073)	(0.077)	(0.083)	(0.086)	
Occupying army conducted misbehavior	0.688	-0.225***	-0.221***	-0.237***	-0.238***	221
		(0.081)	(0.081)	(0.089)	(0.090)	
Number of individuals across all survey responses						472
Controls			Υ		Υ	
Sample	Full	Full	Full	$100 \mathrm{km}$	$100 \mathrm{km}$	Full

 Table A3:
 Allied occupation - survey evidence

Notes: All dependent variables coded as binary indicators. Control variables are sex of the respondent, age, education, indicator for own or parent's experiences. The sample is restricted to individuals for whom the indicated identity of the occupying army matched the May 1945 location. Sample size differs as those indicating "not sure" are excluded in the analysis. Robust standard errors in parentheses. *** p<0.01, ** p<0.05, * p<0.1

Source: Own survey

	State of residence October 29, 1946				
	Brandenburg	Mecklenburg-	Saxony	Saxony-	Thuringia
		Vorpommern		Anhalt	
Place of residence September 1, 1939	(1)	(2)	(3)	(4)	(5)
Soviet Occupied Zone					
Brandenburg	0.710	0.007	0.003	0.009	0.005
Mecklenburg-Vorpommern	0.002	0.533	0.001	0.001	0.001
Saxony	0.003	0.002	0.855	0.008	0.010
Saxony Anhalt	0.003	0.001	0.002	0.720	0.006
Turingia	0.001	0.001	0.002	0.005	0.739
Berlin	0.053	0.012	0.005	0.018	0.015
Trizone	0.009	0.018	0.008	0.020	0.027
Former eastern territories of Germany					
East Prussia	0.021	0.088	0.014	0.023	0.024
Pommerania (east of river Oder)	0.028	0.159	0.003	0.013	0.008
Brandenburg (east of rivers Oder and Lusatian Neisse)	0.059	0.014	0.001	0.008	0.003
Silesia	0.048	0.027	0.075	0.062	0.066
Foreign countries					
Poland	0.027	0.024	0.004	0.019	0.008
Czechoslovakia	0.016	0.080	0.020	0.076	0.070
Soviet Union	0.001	0.001	0.000	0.001	0.000
Other foreign countries	0.013	0.032	0.005	0.015	0.013
Other	0.005	0.001	0.002	0.001	0.004

 Table A4: Population share by pre- and post-war place of residence in East Germany

Notes: Table shows state of residence on October 20, 1946 by place of residence on September 1, 1939. For children born after September 1, 1939, the place of the residence of the parents is being used. Trizone refers to the American, British and French occupied zones. *Other* denotes that the 1939 location was not specified. Source: Census of the Soviet occupation zone from 1946.

	Share that lived in same state						
Mean of dep. variable	0.73	0.74	0.74	0.74	0.74	0.74	
	(1)	(2)	(3)	(4)	(5)	(6)	
Allied	0.021	-0.001	-0.003	-0.039	-0.012	-0.001	
	(0.018)	(0.029)	(0.028)	(0.058)	(0.041)	(0.038)	
Forcing variable $f(\mathbf{g}_i)$			Linear	$Linear \times Allied$			
Border segment FEs					Y	Y	
Controls						Υ	
Bandwidth	Full	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	
Observations	178	89	89	89	89	89	

Table A5: Allied exposure and migration

Notes: Units of observation are counties of the GDR of 1946. The dependent variable is the population share that lived in the same state on October 29, 1946 as on September 1, 1939. Source: Census of the German Occupied Zone from 1946. Border segment FEs consist of four equally large segments cut along the latitude of districts. Controls are distance to Berlin, distance to the inner German border and latitude / longitude. Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1

	1990 vote share				te share	
	1989 protests		Pl	DS	Allianz	
Mean of dep. variable	0.20 (1)	0.20 (2)	0.11 (3)	$0.11 \\ (4)$	0.55(5)	$0.55 \\ (6)$
Allied	0.058 (0.059) [0.052]	$\begin{array}{c} 0.072 \\ (0.061) \\ [0.056] \end{array}$	-0.004 (0.013) [0.011]	-0.002 (0.005) [0.005]	$\begin{array}{c} 0.025 \\ (0.036) \\ [0.026] \end{array}$	$\begin{array}{c} 0.013 \\ (0.014) \\ [0.012] \end{array}$
Forcing variable $f(\mathbf{g}_i)$	$Linear \times Allied$					
Border segment FEs Full controls Sample	$\begin{array}{c} Y\\ Y\\ Population \geq 1,000 \end{array}$		Y Y Full			Y Y
Bandwidth Observations	50km 900	50km 900	50km 2,262	$\begin{array}{c} 50 \mathrm{km} \\ 2,262 \end{array}$	$\begin{array}{c} 50 \mathrm{km} \\ 2,262 \end{array}$	$50 \mathrm{km}$ $2,262$

 Table A6:
 1989 protests and 1990 East German general election

Notes: Units of observation are East German municipalities. Columns 1-2 restrict the sample to municipalities with a population $\geq 1,000$ to make the sample comparable to the 1953 protest estimates in Table 3. Regressions in columns 3-6 are weighted by votes per municipality. PDS was the successor party of the SED, Allianz für Deutschland was an opposition coalition consisting of the parties CDU, DA and DSU. Standard errors clustered at the county level in parentheses. Conley standard errors in square brackets. *** p<0.01, ** p<0.05, * p<0.1

	RIAS sig	anal strength	Dependent variable: Presence of protest					
Mean of dep. variable	70.22	70.22	0.22	0.22	0.22	0.22	0.22	0.22
Ĩ	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Allied	0.530	0.250	0.142***	0.144***	0.169***	0.159***	0.155**	0.154**
	(1.151)	(0.857)	(0.050)	(0.050)	(0.061)	(0.060)	(0.061)	(0.061)
	[0.651]	[0.603]	[0.057]	[0.057]	[0.061]	[0.061]	[0.062]	[0.062]
RIAS signal strength			0.003	0.015	-0.019			-0.020
			(0.003)	(0.021)	(0.022)			(0.023)
			[0.003]	[0.023]	[0.026]			[0.025]
RIAS signal strength ²				-0.000	0.000			0.000
				(0.000)	(0.000)			(0.000)
				[0.000]	[0.000]			[0.000]
Signal strength (alt.)						0.001^{*}	0.005	0.004
						(0.001)	(0.005)	(0.005)
						[0.001]	[0.005]	[0.005]
Signal strength (alt.) ²							-0.000	-0.000
							(0.000)	(0.000)
							[0.000]	[0.000]
Distance to Berlin (km)		-0.078***			-0.003**	-0.002**	-0.002**	-0.003**
		(0.028)			(0.001)	(0.001)	(0.001)	(0.001)
		[0.016]			[0.002]	[0.001]	[0.001]	[0.002]
Forcing variable $f(\mathbf{g}_i)$			Linear × Allied					
Border segment FEs	Y	Y	Y	Y	Y	Y	Y	Y
Full controls		Υ			Υ	Υ	Υ	Υ
Sample			Pop. $\ge 1,000$					
Bandwidth	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$	$50 \mathrm{km}$
Observations	968	968	968	968	968	968	968	968

Notes: Units of observation are all East German municipalities with a population above 1,000. The dependent variable in columns 1-2 is the signal strength of the Radio in the American Sector (RIAS) and in columns 3-8 an indicator for the occurrence of protests during the 1953 uprising. Protest data and RIAS signal strength at the municipality level from Crabtree et al. (2018a). The alternative signal strength indicator stems from 222 cities in East Germany and are merged by nearest neighbour matching. Controls are all municipality-level variables of Table 1 (Panel B), and latitude / longitude. Standard errors clustered at the county-level in parentheses. Conley standard errors in square brackets. *** p<0.01, ** p<0.05, * p<0.1
Dependent variable:	SED vote share					Turnout	Invalid
Mean of dep. var.	0.428(1)	0.428 (2)	$0.394 \\ (3)$	$0.394 \\ (4)$	0.394 (5)	0.937 (6)	0.076 (7)
Allied	-0.032 (0.024) [0.022]	-0.049** (0.025) [0.023]	-0.036 (0.023) [0.021]	-0.028^{**} (0.014) [0.012]	-0.030^{**} (0.013) [0.011]	-0.016** (0.007) [0.006]	-0.004 (0.004) [0.003]
Forcing variable $f(\mathbf{g}_i)$	Linear			$Linear \times Allied$			
Border segment FEs				Y	Y	Y	Y
Full controls					Y	Υ	Υ
Sample	Pop. $\geq 1,000$		All municipal			ities	
Bandwidth Observations	$50 \mathrm{km}$ 1,168	$50 \mathrm{km}$ 1,168	50km 5,649	$50 \mathrm{km}$ 5,649	$50 \mathrm{km}$ 5,649	$50 \mathrm{km}$ 5,649	$50 \mathrm{km}$ 5,649

Table A8: Outcome of the 1946 State elections around the Line of Contact

Notes: Units of observation are East German municipalities in 1946. Dependent variable in columns 1-5 is the vote share of the Socialist Unity Party of Germany (SED) in the 1946 State elections in the Soviet occupation zone. Turnout (column 6) is the share of cast votes (valid and invalid) and Invalid (column 7) denotes the share of invalid votes. All outcomes calculated as shares of the eligible population. Border segment FEs correspond to ten equally large segments of the distribution of latitudes of municipalities in the 50 km bandwidth around the line of contact. Controls are latitude, longitude and all municipality-level variables of Table 1 (Panel B). Regressions are weighted by the eligible voting population. Standard errors clustered at the county level in parentheses. Conley standard errors in square brackets. *** p<0.01, ** p<0.05, * p<0.1

C Retrospective Survey

Survey design and implementation

To complement our findings and to gain a deeper understanding of the mechanisms at play, we ran our own survey in spring 2020 using a professional survey company. The company keeps a large pool of pre-selected respondents who have agreed to be contacted for research purposes.

Targeting individuals among the cohorts who experienced the end of World War II in 2020 is challenging due to attrition and the lower willingness of older individuals to participate in voluntary pools for online surveys. We thus designed the survey to both include first-hand and second-hand accounts of the experience in 1945. We do so by restricting the targeted population to individuals who are aged 60 or older in 2020. To ensure we identify individuals who either have first-hand or second-hand accounts of the occupation in East Germany, we further restrict the sample to those who either themselves lived in East Germany in 1945, or whose parents lived in East Germany in 1945. The main parent we focus on for second-hand accounts is the mother, as fathers were often absent due to combat-related activities. To maximize the sample, we targeted the entire pool of available respondents in the appropriate cohorts. The survey collected basic demographics, their location at the end of the war and recollections of the occupation period in 1945. The survey was designed to be short, taking about 16 minutes to complete on average.

Appendix Table A2 shows basic socio-demographic information. For comparison, we also report the same average characteristics for a representative sample of individuals older than 60 living in East Germany based on the German Socioe-conomic Panel (Goebel et al., 2019). Our survey population is slightly younger as the relevant East German population and, as expected given the targeting, on average higher educated. The distribution of states of residence matches reasonably well. Despite our attempts to maximize the sample of first-hand accounts, the majority of respondents are second hand accounts. In our final sample, 68% of the observations stem from second-hand accounts from the mothers, 28% from fathers and only 3% from first hand accounts. Our results are robust to holding constant these differences using respondent type fixed effects.

A second challenge to such a retrospective survey is recall bias. Recollections from a period of upheaval may be particularly blurry, and second-hand accounts may be less reliable due to parents' reluctance to share often traumatic experiences. In our context, there are two particular challenges: first, respondents may not correctly recall the initial occupation of the Allied forces due to the short-lived nature. We thus cross-check the responses by geolocating the reported residence at the end of the war. The final sample only included

individuals who provided accurate responses pertaining the initial occupying force's identity (66%). Second, respondents may not be certain about specific details. To capture this, the survey explicitly allowed respondents to state that they are unsure about certain events. We exclude individuals who provide responses that they are uncertain about.⁵² We follow the same procedure when we analyze the change of occupying army, i.e. we restrict the sample to individuals who at that time lived in Allied occupied territory and correctly identify the Red Army as the new occupying army (see Appendix Figure A11).

Full wording of questions (translated)

Survey questions shown in Figure 6

Panel a:

- How was the arrival of the occupying army perceived by the population in your [or your parent's] place of residence?
 - (1) Very positive (2) A bit positive (3) Neutral (4) A bit negative (5) Very negative (6) Not sure
 - * Coded as positive if very positive or a bit positive is indicated
- Did the occupying army distribute food?
 - (1) Yes (2) No (3) Not sure
- Did the occupying army distribute medicine / care for the injured?

- (1) Yes (2) No (3) Not sure

- Did the occupying army mediate conflicts / enforce justice?
 - (1) Yes (2) No (3) Not sure
- Did the occupying army conduct misbehavior?
 - (1) Yes (2) No (3) Not sure

Panel b:

- How was the change of the occupying army perceived?
 - (1) Very positive (2) A bit positive (3) Neutral (4) A bit negative (5) Very negative (6) Not sure

 $^{^{52}}$ We do not find that the propensity to be uncertain varies significantly across the line of contact. The results are robust to coding unsure responses as negative responses.

* Coded as negative if *very negative* or *a bit negative* is indicated

• Did the new occupying army distributed food?

- (1) Less than before (2) More than before (3) Not sure

• Did the new occupying army distributed medicine / care for the injured?

- (1) Less than before (2) More than before (3) Not sure

- Did the new occupying army mediate conflicts / enforce justice?
 - (1) Less than before (2) More than before (3) Not sure
- Did the new occupying army conduct misbehavior?
 - (1) Less than before (2) More than before (3) Not sure